

## ORIGINAL ARTICLE/ARTICLE ORIGINAL

# Racial Integration and the Problem of Relational Devaluation

D. C. Matthew (D)



#### Abstract

This article argues that blacks should reject integration on self-protective and solidarity grounds. It distinguishes two aspects of black devaluation: a 'stigmatization' aspect that has to do with the fact that blacks are subject to various forms of discrimination, and an aesthetic aspect ('phenotypic devaluation') that concerns the aesthetic devaluation of characteristically black phenotypic traits. It identifies four self-worth harms that integration may inflict, and suggests that these may outweigh the benefits of integration. Further, it argues that, while the integrating process may reduce stigmatization, there is less reason to think that it can do the same for phenotypic devaluation.

#### Résumé

Cet article soutient que les noirs devraient rejeter l'intégration pour des raisons d'autoprotection et de solidarité. Il distingue deux aspects de la dévalorisation des noirs : un aspect de « stigmatisation » qui a à voir avec le fait que les noirs sont soumis à diverses formes de discrimination, et un aspect esthétique (« dévalorisation phénotypique ») qui concerne la dévalorisation esthétique de traits phénotypiques typiquement noirs. Il identifie quatre préjudices à l'estime de soi que l'intégration peut infliger et suggère que ceux-ci peuvent l'emporter sur les avantages de l'intégration. En outre, il soutient que si le processus d'intégration peut réduire la stigmatisation, il y a moins de raisons de penser que ce processus puisse faire de même pour la dévalorisation phénotypique.

Keywords: racial integration; relational value; self-worth; assimilation; black solidarity

In recent years, a number of commentators have issued renewed calls for racial integration (e.g., Adams, 2006; Anderson, 2010; Cashin, 2004; Patterson, 1997, esp. pp. 171–203; Stanley, 2017). Their arguments usually involve a descriptive aspect in which segregation is identified as a cause of one or more social problems, and a prescriptive aspect in which integration is proposed as the solution. Michelle

<sup>©</sup> The Author(s), 2021. Published by Cambridge University Press on behalf of the Canadian Philosophical Association/Publié par Cambridge University Press au nom de l'Association canadienne de philosophie.



<sup>&</sup>lt;sup>1</sup> I should say something about Stanley's (2017) inclusion on this list. Although, all things considered, I think it is appropriate to group her among integrationists, it should be acknowledged that she is in some ways much closer to my position. She argues that, while a worthwhile model of integration can be found, there is reason to doubt whether it can be fully implemented, so that what she calls "black integration pessimism" is justified.

Adams (2006), for instance, claims that segregation "structures, maintains, and perpetuates inequality across virtually every indicia of social, political, educational and economic well being" (p. 275), and Elizabeth Anderson argues that segregation not only causes unjust group inequality, but expresses stigmatizing ideas, engenders stigmatizing stereotypes and discrimination, and even undermines democracy (Anderson, 2010, Chs. 2–3). Having identified segregation as a cause of these problems, these 'new integrationists' go on to propose integration — especially residential integration — as the solution.<sup>2</sup> As Anderson (2010) writes, "If racial segregation is the problem, it stands to reason that racial integration is the remedy" (p. 112). In this paper I will argue that these new integrationists are wrong. Even if segregation is a problem, integration may not be the solution. Integration, I will argue, threatens black self-worth, and for blacks a concern to safeguard their self-worth should outweigh the justice and other benefits that integration is supposed to bring.

Although this paper opposes integration in terms of values that are race-neutral, it focuses exclusively on the case of blacks.3 There are good reasons for doing so. As some have argued, the emerging pattern of race relations in the United States, Canada, and elsewhere appears to be one in which the main division is not between whites and non-whites, but between blacks and non-blacks. For example, George Yancey (2003) has argued that, like the barriers to the assimilation of European immigrant groups before them, barriers to Asian American and Hispanic American assimilation are receding and will continue to recede over time, thereby allowing these groups to join the mainstream. But the same, he insists, is not true for African Americans, who he believes are "destined to remain an outcast race" (p. 13). And in their research on what they call "the new diversity" in the US, Jennifer Lee and Frank Bean (2010) find a consistent pattern of "black exceptionalism" in the marriage market and multiracial identification, suggesting, they say, that "cultural boundaries" are dissolving more quickly for new immigrant groups like Asians and Hispanics than for blacks (see also Alba et al., 2018; Gans, 1999; Sears, 2015; Warren & Twine, 1997; for Canada, see Kymlicka, 1998, Ch. 5).

I believe that these facts reflect what Roy Brooks (1996) describes as the "devaluation of the Negroid phenotype" (p. 205). One aspect of this devaluation surely concerns the fact that the phenotypic traits that historically have in part defined blackness — darker skin, broader or flatter noses, etc. — have been ascribed certain meanings that have been used as the basis for holding blacks to be of lesser worth and for justifying various forms of discrimination against them. At the same time, however, many seem to regard these traits as contributing to a person's attractiveness, with characteristically white phenotypic traits, for example, thought to be more attractive than characteristically black ones.

<sup>&</sup>lt;sup>2</sup> I borrow the term "new integrationists" from Shelby (2016, Ch. 2), who uses it to distinguish these newer advocates from those of the civil rights era who wanted to end Jim Crow prohibitions and various forms of discrimination (e.g., employment discrimination).

<sup>&</sup>lt;sup>3</sup> In this paper, when I speak of 'African Americans' I will specifically have Black Americans in mind, whereas I use 'blacks' to refer to persons of African descent, regardless of nationality. So, as used here, 'blacks' includes both African Americans and other blacks. I adopt this terminology in part because, while much of the literature cited below focuses on the situation of African Americans, I intend my argument to have application for other blacks, as I go on to explain.

This raises the prospect that there may be two aspects of black devaluation: what we may call a 'stigmatization' aspect, and an aesthetic aspect. The stigmatization aspect has to do with the fact that blacks are subject to negative stereotypes, interpersonal rejection, and various forms of discrimination because of their race (on stigma, see Crocker et al., 1998; Goffman, 1963; see also Loury, 2002). The aesthetic aspect concerns the widely shared judgement that characteristically black phenotypic traits (e.g., dark skin, etc.) are less attractive than non-black ones, with the result that many consider blacks to be singularly unattractive. I call this aesthetic-based devaluation *phenotypic devaluation*.

I will argue that stigmatization and phenotypic devaluation tend to depress the 'relational value' of blacks, or how much blacks are affectively (or non-instrumentally) valued by others. Of course, both phenotypic devaluation and especially stigmatization produce other harms that have little or nothing to do with relational value. I focus on the harms of devaluation because I believe that such harms may endure even in an integrated society. I understand the harms in question in terms of self-worth, and identify four of them: one having to do with self-esteem, another with self-respect, a third with feelings of inferiority, and the fourth with the acceptance of inferiority.

Although this paper has much to say about the problem of stigmatization, it is ultimately more interested in the problem of phenotypic devaluation. The problem of stigmatization is, to be sure, a very serious and urgent one: it contributes to — and militates against reforms in — biased policing, injustice in the criminal justice system (including mass incarceration), and inequalities in various other domains (e.g., education and health care). In my view, while one cannot plausibly deny that there has been remarkable progress in racial attitudes over the past half-century, it would be a mistake to assume that we are on an inexorable march of progress. What progress there has been has at times been halting, reverses have not been unknown, and there are signs that more reversals are possible.

But this paper was written with an eye toward the future, including the distant future, and it is hard to be confident about what will happen over the long term. It may be that we will eventually solve the stigmatization problem. But I argue that the decline of stigmatization may put into sharper relief a different problem (phenotypic devaluation) with which it has always co-existed — and one that produces similar self-worth harms.

If the stigmatization problem cannot be solved, that strengthens the argument, for the two problems together are bigger threats to black self-worth than the problem of phenotypic devaluation alone. Indeed, one might reject integration for purely stigma-related reasons. But such arguments invite a particular kind of reply: that the very integrating process itself will reduce stigmatization. For example, one leading theory of intergroup relations holds that contact between majority group members and minority group members can reduce 'prejudice,' especially if certain conditions

<sup>&</sup>lt;sup>4</sup> I do not attempt to establish this; likely readers will recognize it as true from their experience of living in the culture. Nor will I challenge it; I will leave aside questions regarding the objectivity of aesthetic judgements. I will say though, as I hope the paper as a whole implicitly shows, that I do not share the judgements that I describe

obtain (Dovidio et al., 2017; Pettigrew & Tropp, 2011). I argue, however, that even if such contact can reduce prejudice, there is less reason to believe that it can significantly reduce phenotypic devaluation.

In the US, integration has largely fallen out of favour among African Americans. In part, this is because of the persistence of segregation. As Adams (2006) has noted, many African Americans reject integration "precisely because segregation continues to be an enduring feature of American life" (p. 287). But in part it also reflects African American desire to retain the identities and institutions developed over the course of their long oppression. Tommie Shelby (2012) has written:

The victims of white domination have often rejected the tenets of racial ideology that demean, misrepresent, and stigmatize them ... But instead of discarding the race idea altogether, many have cultivated positive racial identities and racial solidarity that have enabled them to survive and to resist their oppression ... [They] are often proud of their collective identities and sometimes seek to celebrate them publicly. On the basis of these identities, they have formed race-based organizations and institutions for political, religious, economic, and educational purposes. (p. 349)

In other words, African Americans have formed their own institutions and forged distinct identities in the face of racial oppression, and as a result many are reluctant to give them up in the name of integration. Some critics, and even some advocates, of integration have made much of this reluctance. For example, in a recent book, Sharon Stanley (2017) is at pains to develop a model of integration that does not demand assimilation into white norms and institutions or the dissolution of black institutions or cultural and social practices.

In my view, however, it may be better not to let our thinking about integration be much influenced by the attachment of blacks to longstanding cultural practices and institutions. Wanting to keep such practices and institutions may be a perfectly good reason to resist integration, but there may be better reasons still. I think that there are such reasons, and I suggest that we may gain a clearer appreciation of them if in considering the case for integration it is assumed that blacks are *willing* to integrate. If the case against integration is strong even in the case of a willing group, the case against it where the group is unwilling must be stronger still. So I will assume that the black population is willing to integrate, and thus will abstract from black attachment to longstanding cultural practices and institutions.

In addition, if integration no longer appeals to many African Americans, this may not be true of other blacks. The question of integration is not just an American one; Black British people, African Canadians, and other blacks must also confront it. This suggests that, in thinking about its merits, it may be better not to restrict our attention to the American context. If we must think of any single context, a better one may be a place like Canada, where most blacks are either recent immigrants or the descendants of recent immigrants, and so should not have the same attachment to longstanding (Canada-shaped) cultural practices and institutions (on blacks in Canada, see

Winks, 1997; see also Mensah, 2010). There is also some reason to think that such immigrants may have a different attitude toward integration.<sup>5</sup>

Though this paper does not have much to say about the concept of race, some readers may wonder if there is not a working conception that is implicitly being employed. Given this, I will note here that I regard its argument as being compatible with the two main positions in the debate about the nature of race: biological racial realism and social constructionism. Biological racial realists hold that race is *biologically* real, typically holding that it is either a natural kind in biology or a real biological kind (Spencer, 2012). Social constructionists, by contrast, hold that race is a *social* construct, a product of our social practices. On this view, race is real, but real as a social kind. I believe that the ideas defended in this paper are compatible with both of these views, and I regard this as a virtue of the account. These views are varieties of racial realism. Against racial realist views are anti-realist views, which deny that race is real in any sense (e.g., Appiah, 1996). My argument may be in greater tension with anti-realist views (see Mallon, 2017). But I believe — though I cannot argue for this here — that realist views have more to be said for them.

Finally, this paper makes two assumptions that should be explicitly noted. First, I will assume that the black group is a *numerical minority* with a non-black majority. There is some reason to think that the assimilative effects of integration will be attenuated if the black population is too large. So I assume that blacks constitute no more than 20% of the population. Second, I will initially assume that the black population is a relatively homogenous group phenotypically, with phenotypes that include those features commonly associated with blacks (dark skin, more tightly coiled hair textures, etc.). Later I will relax this assumption and consider the phenotypic variation that in fact characterizes black people.

The paper proceeds as follows. Section I further describes phenotypic devaluation, distinguishes it from racism, and explains one way that it raises a problem of justice. Section II presents the self-esteem argument, and Section III explains the connection between integration and assimilation. The following section, Section IV, discusses the self-respect-based argument and the feelings/relations of inferiority arguments. The claimed prejudice-reduction benefits of integration are examined in Section V. This section also considers objections as well as the difficulty that phenotypic variation presents for my argument. Section VI provides a conclusion.

<sup>&</sup>lt;sup>5</sup> For an account that suggests that blacks with differing immigration histories should have different attitudes toward assimilation, see (Ogbu & Simon, 1998).

<sup>&</sup>lt;sup>6</sup> Some may want to challenge one or both of these claims: denying that my arguments really are compatible with both biological racial realism and social constructionism, and/or questioning whether this should really be considered a virtue. I will not take up either objection here. I will say though that I believe that it is a mistake to believe, as many do, that there is no case to be made for thinking of races as biologically real. For examples of how this may be done, see (Kitchner, 2007; Spencer, 2019).

<sup>&</sup>lt;sup>7</sup>I add here that if we find the 'race' idea too problematic, some suggested alternatives are also compatible with the arguments developed here. This is true, for example, of Lionel McPherson's (2015) 'socioancestral' groups.

<sup>&</sup>lt;sup>8</sup> To be sure, there is significant phenotypic variation among black subpopulation groups (compare, for example, most West African groups with some East African groups). This fact is not lost on me, nor is the way in which it may limit the scope of my argument.

I.

Phenotypic devaluation, I have said, is the relative aesthetic devaluation of characteristically black phenotypic traits. In this section, I explain why it should be distinguished from racism, but may nevertheless give rise to racial injustice.

Before proceeding any further, we should be a little clearer about what phenotypic devaluation involves. Three points are important to emphasize. First, it can involve not just the mere mental devaluation of certain phenotypic traits, but the external expression of this. In fact, I will be especially interested in its external expression in interpersonal relations. All else being equal, people not only prefer to be around those who they consider physically attractive, they also prefer not to be around those who they consider sufficiently unattractive. Moreover, they are likely to treat the former better than the latter. Phenotypic devaluation can of course be expressed in other media (e.g., movies and television) as well, and when this is the case, it will have more harmful effects than when it is expressed in interpersonal relations alone. Admittedly, some of the research I cite below relies on these other expressions of it, but I will assume, perhaps unrealistically, that there are relatively easy measures that can be adopted to significantly reduce them. In the case of interpersonal relations, however, its expression — where the underlying aesthetic judgements exist — is ineliminable short of the elimination of interpersonal relations itself.

Second, phenotypic devaluation can be a purely aesthetic devaluation, but some expressions of it are not 'pure' because the aesthetic judgements involved are not 'pure.' Impure aesthetic judgements can be of different kinds. One such kind is when the group identity of the person being judged negatively affects ostensibly aesthetic judgements of bodily and facial beauty. If I have a negative attitude toward Asians, for example, this very fact can affect my judgements of phenotypic traits that Asians are more likely to possess. Although much current phenotypic devaluation may involve such 'impurities,' I assume that it can be free of them. I'll return to this issue below.

Finally, note that what is most significant about phenotypic devaluation is not that some people are judged to be unattractive, but that some people are judged to be unattractive below a certain minimum. It is of course a commonplace observation that some individuals are considered more attractive than others, and this is no different if we restrict our attention to blacks. But what is significant about phenotypic devaluation is the possibility that many blacks may fall below a threshold the satisfaction of which secures a person a sufficient degree of relational value.

Now the history of anti-black racism, it is clear, is bound up with the reactions and meanings attributed to phenotypic traits such as skin colour and nose shape. Nevertheless, it is possible to imagine progress in race relations so dramatic that these reactions and meanings are eliminated. My claim is that the aesthetic judgements that in part constitute phenotypic devaluation may still remain. In this scenario, no black person need ever be hated, held to be of less moral worth, judged

<sup>&</sup>lt;sup>9</sup> The distinction that I am drawing here is similar to one that Xiaofei Liu (2018) draws between what he calls "simple lookism" and "loaded lookism" (pp. 277–278).

<sup>&</sup>lt;sup>10</sup> Perhaps something like what Richard Wasserstrom (1977) calls "the assimilation ideal" (p. 604) is realized, except that certain phenotypic traits are important bases of differential aesthetic judgements and the treatments that follow from them

less intelligent, less law-abiding, etc. because of race. If this is so, phenotypic devaluation should be distinguished from racism.<sup>11</sup>

To be sure, in such a scenario, phenotypic devaluation would still lead to a kind of inequality of *esteem* between individuals who tend to fall into familiar racial groups. Persons with lighter skin, certain nose shapes, hair textures, and so on may be judged to be more physically attractive than persons with darker skin, more tightly coiled hair, etc., and the persons judged more favourably for these characteristics would be more relationally valued as a result. These facts may lead some to balk at my suggestion that phenotypic devaluation should be distinguished from racism. In a paper on the problem that inequalities of esteem present for relational egalitarians, Carina Fourie (2015) writes:

Besides the practical objection that in reality the only people who would esteem white skin seem indeed to be motivated by racism ... this form of esteem seems worrying. We can categorize this concern as being about the grounds for esteem. Certain kinds of reasons for esteem would be objectionable from a socially egalitarian perspective ... [I]n a world characterized by racial oppression and discrimination, "esteeming" race seems far too close to a violation of respect for comfort. Esteeming people because of their skin color does not seem to gel with the notion that we should treat people as equals as it seems strongly associated with discrimination and with treating certain races as inferior and superior, even if, as a "genuine" case of esteeming, no injustice is really indicated. (p. 98; see also Runciman, 1967)

But this argument may be faulted on a number of grounds. First, and as suggested, we might have a world in which characteristically white features are esteemed and characteristically black features are disesteemed that is *not* characterized by racial oppression. To the extent that "esteeming race" appears troubling because it takes place in a world characterized by racial oppression — and arguably what is being esteemed is not race per se, but certain race-linked phenotypic traits — we can simply imagine a world not so characterized, and the troubling appearance should disappear (Cf. Fourie, 2015, pp. 98–99). This, to be sure, is not to say that such a world could not *itself* be oppressive in any way. After all, inequality of race-linked esteem can certainly make individuals' lives miserable. My point is simply that a world where phenotypic devaluation is widespread may not be one that is otherwise oppressive (and so be troubling in some way because of that).

Second, even if inequalities of esteem are "strongly associated with discrimination" (Fourie, 2015, p. 98), this wouldn't make them racist. Though differential treatment of persons because of phenotypic traits can properly be spoken of in the language of 'discrimination,' we should not assume that discrimination — even racial discrimination — must in some way involve racism. While it is possible to defend a behavioural conception of racism, so that 'racism' is primarily attributed to actions and acts of racial discrimination always come out as 'racist' (see Flew, 1990; Philips, 1984), this

<sup>&</sup>lt;sup>11</sup> Now some will challenge this by claiming that such racially differentiated aesthetic judgements depend on racism, so that in its absence they could not in fact remain. I discuss this objection in Section V below.

sort of view should be rejected. It is better if we distinguish between *racial* discrimination and *racist* discrimination, so that while every instance of the latter is an instance of the former, the reverse is not true: not every case of racial discrimination need be a case of racist discrimination (see Matthew, 2017; see also Garcia, 1996, pp. 16, 28). If a black man looking for a wife refuses to consider any woman who is not black, he is surely discriminating by race in his search. But it is another question whether this discrimination is racist. Whether an act is racist seems to depend in some way on the mental state of the person performing it, but it is not clear if the same is true, in the same way, for whether an act is racially discriminatory. <sup>14</sup>

As for Fourie's claim that esteeming phenotypic traits unequally distributed by race is "associated" with "treating certain races as inferior and superior," there is a sense in which this is true, but that sense does not justify the claim that the treatment is racist. For to the extent that persons with different phenotypic traits are unequally esteemed because of this, they will tend to be treated unequally, and this unequal treatment will of necessity produce an inequality between the preferred and the dispreferred. One might choose to describe this result in terms of treating some races as "inferior" and others as "superior." But given that blackness is not associated with anything like (overt or covert) anti-black animus or inferiorizing beliefs in the society we are imagining, it is not clear that we should think of this in terms of 'racism.'

To be clear, even if it is true that a society where phenotypic devaluation is wide-spread need not be either racist or otherwise racially oppressive, this need not put it in the clear as far as race is concerned. Being 'racist' or 'racially oppressive' are simply two dimensions along which something may be evaluated in the racial domain. There may be other ways that thing — whether an act, a feeling, a situation, etc. — offends against ideals in this domain (see Matthew, 2017; see also Blum, 2002, Chs. 1 and 3). Thus if a society characterized by inequality of phenotype-based esteem is one in which it can truly be said that some races are treated as inferior and some as superior, this fact may be of normative significance from a race-related point of view, even if it is not racist. It may also, of course, be of normative significance from a non-race-related point of view. For example, it may be troubling from a relational egalitarian perspective, as Fourie suggests. <sup>15</sup>

<sup>&</sup>lt;sup>12</sup> I use 'discrimination' in a descriptive or non-moralized sense, which means that saying that an act is an act of discrimination does not carry any final evaluative judgement concerning it; descriptive conceptions are evaluatively non-committed. So, though the man in the example discriminates, it is a further question not only whether this discrimination is racist, but also whether it is wrong.

<sup>&</sup>lt;sup>13</sup> Note that I disclaim any advantage that may attend this argument owing to the fact that it is a *black* man who wants a black woman to be his wife; suitably clarified (to exclude certain objectionable cases), a similar example of a white person wanting a white partner should do as well. (I am not, however, claiming that the cases are entirely symmetrical.) I reject frequently made claims that blacks and other people of colour can't be racist. Anybody, in my view, can be racist, but all racism may not be equal. For a defence of this view, see (Blum, 2002, Ch. 2).

<sup>&</sup>lt;sup>14</sup> I should say here that I reject accounts of racism (including some accounts of institutional racism) that make its existence turn on the consequences for victims. J. L. A. Garcia has produced an illuminating body of work on this issue (see esp. Garcia, 1996; see also Garcia, 1997).

<sup>&</sup>lt;sup>15</sup> In fact, relationships might offend against relational egalitarian ideals for either race-related reasons or non-race-related reasons. On relational egalitarianism, see, e.g., (Anderson, 1999; Fourie, 2012; Scheffler, 2015). I will come back to Fourie's point about treating some races as inferior.

So even if not racist, race-linked, phenotype-based inequalities of esteem may be objectionable for other race-related reasons. One possible argument here deserves mention. It may be suggested that given that characteristically black phenotypes have in the past been judged to be markers of inferiority, negative aesthetic judgements of them will express stigmatizing messages and so be disrespectful to blacks (Mitchell & Wells, 2018). It might be claimed that such stigmatizing messages would carry over even in our imagined society. Now though I do not want to deny that, given our history, such negative judgements may express stigmatizing messages, I do not think that any such messages will persist indefinitely, even if they won't fade away immediately. Here we need to think on an appropriately long-term basis (in terms of hundreds of years, and not simply years or even decades), and recognize that while the past can powerfully affect the future, it is not infinitely powerful.<sup>16</sup>

I have argued that it is not the case that a society characterized by unequal phenotypic esteem need be one that is characterized by racism. But now I want to argue — consistent with the above analysis — that it may still be a racially unjust society. Even if not themselves unjust, inequalities of esteem might still harm those who are disfavoured, and such harms might figure in a theory of justice (even apart from relational egalitarian considerations, which are often regarded as considerations of justice). For example, they may figure in virtue of the opportunity (formal or substantive) costs that they involve. Thus research has shown that persons who are judged to be more attractive are more likely to be voted for, hired, and to receive favourable outcomes in court (Frevert & Walker, 2014). All of this reduces the opportunities (to be elected, hired, and to receive a favourable judicial outcome) of those judged to be less attractive. Unequal phenotypic esteem may also lead to injustice through social capital effects: given that information about jobs is often provided informally through social networks, as Anderson (2010) emphasizes, racially segregated social networks can disadvantage members of excluded groups by cutting them off from such information (pp. 33-34; see also Roithmayr, 2014, Ch. 6).

But suppose that this line of thinking is wrong. Suppose that inequality of phenotypic esteem shouldn't be regarded as giving rise to injustice (or otherwise being wrong). I maintain that phenotypic devaluation may still pose a significant obstacle to integration. For whether or not it gives rise to injustice, being relationally devalued because of one's phenotypic traits may still undermine one's self-worth. From the perspective of those who are so devalued, it should not matter whether the devalued treatment to which they are subject is morally in the clear. They may have prudential and perhaps even moral reasons to resist being in situations that threaten to compromise their self-worth.

To show that phenotypic devaluation should be distinguished from racism, I have imagined a society that is free from some of the reactions and social meanings that dark skin, etc. have tended to attract. I want now to come back to the real world, where these reactions and meanings are very much still with us. Blacks remain a highly stigmatized group, and in what follows I do not abstract away from this fact. The main point of this section was to suggest that it may be possible for anti-

 $<sup>^{16}</sup>$  I thank some of the anonymous referees for getting me to see the inadequacy of my initial treatment of the issues just discussed.

black racism to decline while phenotypic devaluation persists — both now exist, and are linked, but this link may not be robust. <sup>17</sup> Now if integration is to be viable at all, its advocates should have to make the case that it will eliminate or at least significantly reduce anti-black racism and discrimination. But I will argue that even if integrationists can make such a case, phenotypic devaluation may still remain. Before I get to these matters though, we need to first understand how integration may intensify the self-worth harms of stigmatization and phenotypic devaluation. I begin with self-esteem.

#### II.

In the most general sense, self-esteem is a form of self-worth. But it is not a coldly evaluative assessment of worth; it is an affectively laden one. As Mark Leary and Roy Baumeister (2000) put it, "At its core, self-esteem refers to how we feel about ourselves" (p. 3). Nevertheless, it does have an evaluative dimension, and psychologists have distinguished distinct sources or bases of the self-evaluations relevant to it (Leary & Terry, 2013). I begin by explaining why two of these appear to suggest that African Americans should have low trait self-esteem. These sources roughly correspond to the sources identified by reflective appraisal theory and social comparison theory — 'roughly' because I simplify considerably, and neither theory, in fact, straightforwardly predicts that African Americans should have poor self-esteem (see Corcoran et al., 2011; Wallace & Tice, 2012).

First, our self-evaluations seem in part to reflect the appraisal of others. The perceptions of others, in other words, seem to influence our self-perceptions, and so in turn our self-views. So by this source of self-evaluation, if we seem in general to be positively appraised by others, we should in turn positively appraise ourselves as well, and similarly for negative appraisals. A second basis for self-evaluation is social comparison. If we want to come to some judgement about ourselves on some particular dimension, one way to come to it is to examine how we fare relative to others. The results of these should then inform our self-views. Given both stigmatization and phenotypic devaluation, it would seem that both of these sources would predict that African Americans should have low self-esteem.

A leading theory of self-esteem also seems to suggest the same. Sociometer theory holds that the self-esteem system is one that "monitors the quality of an individual's interpersonal relationships and motivates behaviors that help the person to maintain a minimum level of acceptance by other people" (Leary & Baumeister, 2000, p. 9; see also Holden et al., 2016, Leary, 2005). Self-esteem, on this view, is essentially a marker of our relational value. Now if self-esteem is a subjective monitor of a person's eligibility for social inclusion, then it should depend on the factors that influence

<sup>&</sup>lt;sup>17</sup> Importantly, the link may not be in the direction that some may imagine: instead of racism determining the course of phenotypic devaluation, phenotypic devaluation may determine the course of racism. Recent research by Igor Ryabov (2019) is consistent with this possibility. He found that, for young adults, physical attractiveness mediates the relationship between skin tone and status attainment, with the result that "stratifications by physical attractiveness and phenotype among blacks overlap to a high degree" (p. 77). More generally, there is some evidence that European aesthetic standards were important to the rise of racism in the modern period; see esp. (West, 2002, Ch. 2).

inclusion. In addition to the standardly identified characteristics, membership in a stigmatized group is one such factor. Stigmatized persons have a 'discredited' or devalued social identity, and this tends to make them less eligible for social inclusion (Goffman, 1963; see also Crocker et al., 1998, pp. 505-506). As for the standardly identified characteristics, they include likeability, competence, trustworthiness and physical attractiveness (Leary & Baumeister, 2000, pp. 17-18; see also Leary & Allen, 2011). As one researcher reports, "self-evaluations in the domain of physical appearance" have been found to be "inextricably linked to global self-esteem" (Harter, 1993, p. 117). The reason is obvious: people prefer to be around, and treat better, those who they regard as physically attractive. Because this is so, persons who are generally judged as attractive should receive more offers of inclusion than persons who are judged as unattractive, and so should tend to have higher self-esteem (Leary & Baumeister, 2000, pp. 17-18). And this is something that should hold true not just in certain special (e.g., dating) contexts, for the attractive are likely to be perceived as more relationally valuable in a way that is not context-dependent. As Leary and Ashley Batts Allen (2001) argue, "even when one of these sets of characteristics [i.e., likeability, competence, etc.] is objectively irrelevant in a particular relational context, it can nonetheless influence other people's perceptions of the person and, thus, his or her relational value" (p. 39). All else being equal, this means that African Americans should generally have a lower relational value than persons who are not African American, and so lower self-esteem. Critical to this is the potentially chronic character of negative feedback about relational value.

So different considerations appear to predict that African Americans should have low self-esteem. In study after study, however, researchers have consistently found that African Americans have self-esteem that is equal to or higher than that of other Americans. In a meta-analysis of well over 300 studies of self-esteem, Jean Twenge and Jennifer Crocker (2002) found that African Americans had higher self-esteem scores than white Americans, who themselves had higher scores than other racial groups (Hispanic Americans, Asian Americans, and American Indians).

What explains African American resilience? Social identity theory points toward at least a partial answer. Positing that people are motivated to enhance and maintain a positive self-concept, and that the positive evaluation of the social groups to which they belong is key to a positive self-concept, social identity theory suggests that individuals from low status groups may strive to achieve a positive social identity (Tajfel & Turner, 2004; see also Ellemers & Haslam, 2012). It claims that when a group's low status is perceived to be illegitimate and the boundaries around it relatively impermeable, its members will attempt to secure a positive social identity using group strategies such as differentiating their group from outgroups, and engaging in social comparisons with relevant outgroups. According to Henri Tajfel and John Turner (2004), key to the use of these strategies is subjective identification with the in-group; for individuals to be able to use them, they write, they "must have internalized their group membership as an aspect of their self-concept: they must be subjectively identified with the relevant in-group" (p. 284). <sup>18</sup>

<sup>&</sup>lt;sup>18</sup> Tajfel and Turner explain that in other circumstances (i.e., when boundaries are perceived to be permeable and subjective identification is low), members of low status groups may pursue more individual

Social identity theory appears to explain African American psychological wellbeing in the face of devaluation. Research by Michael Hughes and colleagues (2015) found that, for African Americans, strongly identifying with their racial group and evaluating it very positively promoted "self-worth, self-efficacy, and good mental health" (p. 42). Social identity theory also seems to explain, at least in part, why African American women are able to maintain positive self-evaluations about their appearance. One particularly noteworthy study, by Jalmeen Makkar and Michael Strube (1995), compared the responses of black women exposed to a white standard of beauty and a black standard of beauty, both in the form of colour photographs of women pre-rated as attractive. They were then asked to complete ratings of their own attractiveness. A black self-consciousness scale was administered to measure the strength of black identity. As Makkar and Strube predicted, following exposure to the white models (but not the black ones) those blacks with a stronger black identity rated their own attractiveness higher, and that of the white models lower, than did those with a weaker black identity. Makkar and Strube suggest that these results "indicate that women who do not embrace their own ethnic identity may be more susceptible to comparison to White standards of beauty, resulting in lower self-acceptance. By contrast, women with high Black identity are relatively less susceptible to the potentially damaging comparison to unrealistic standards of beauty offered by the White culture" (Makkar & Strube, 1995, p. 1560).

So it seems that a positive group identity can buffer some of the negative feedback of others and so protect black psychological well-being. Some researchers have suggested specific psychological strategies that members of stigmatized groups might use. Perhaps the best-known work of this kind is that of Jennifer Crocker and Brenda Major (1989). They suggest that "the more an individual has structured his or her self-concept around membership in a group that is devalued, deprived, or discriminated against, the better that individual feels about him or herself in terms of global self-esteem ... [for] identification with the stigmatized group allows the individual to use [certain] group-level self-protective strategies" (p. 620; see also Major et al., 2003). 19 They identify three such strategies: i) attributing negative outcomes to group membership (instead of stable, internal personal attributes); ii) devaluing those attributes or dimensions on which the members of the group fare poorly, relative to others; and iii) using members of the ingroup for comparison purposes, which Crocker and Major (1989) say "allow[s] the stigmatized to avoid the selfesteem threatening consequences of outgroup social comparisons" (p. 614). In each case, integration — if it leads to assimilation, as I will argue it will — should reduce the ability and/or willingness of blacks to employ these strategies. Take the devaluing strategy, for example. It will presumably be more difficult for the members

<sup>(</sup>rather than group) strategies (e.g., leaving the group), which they claim is destructive of group solidarity and leaves the group's low status intact.

<sup>&</sup>lt;sup>19</sup> This 'stigma as self-protection' theory can be interpreted as a stand-alone theory such that what is key is the mere membership in a stigmatized group (see Twenge & Crocker, 2002). This is not how I will understand it, however. Instead, I hold that it is having a positive and central racial identity — which all stigmatized racial groups need not have — that protects black self-esteem by facilitating the use of the self-protective strategies that I go on to describe (Twenge & Crocker recognize this possibility (pp. 373, 387)).

of a group to devalue an attribute or dimension valued highly by the larger culture if it is too immersed in that larger culture.

Once its implications are clarified, sociometer theory is also able to explain why African Americans do not have low self-esteem. As Leary and Baumeister (2000) explain, our need for belonging is not boundless: "people need only a certain level of belongingness and are not motivated to be valued and included by everyone. Once the person's need to belong has been fulfilled by relationships with some relatively circumscribed group of people, relational devaluation — even outright rejection — by others may have little effect on self-esteem" (p. 37). But once again, there is reason here to doubt whether the circumstances that protect black self-esteem will carry over to an integrated society, for Leary and Baumeister go on to suggest that "members of discriminated-against groups should suffer a decrement in self-esteem only to the extent that they either do not otherwise have an adequate social network or desire to be accepted by the out-group members who reject them" (p. 37; see also Leary, 2005). Since in an integrated society blacks would presumably not place racial restrictions on those seen as eligible to fulfill their belonging needs, sociometer theory also seems to suggest that they should have lower self-esteem in such a society.

So it seems that black self-esteem would be better protected in a segregated setting than in an integrated one. This explains one little-known fact about segregation in the Jim Crow past. In coming to its decision in its landmark ruling striking down the *de jure* segregation of public schools, *Brown v. Board of Education of Topeka* (1954), the US Supreme Court famously cited some doll studies by psychologists Kenneth and Mamie Clark. In a series of studies, the Clarks presented black children between the ages of 3 and 7 with otherwise identical white and brown dolls and asked them, among other things, to indicate which doll they wanted to play with and which doll they thought looked "bad" (Clark & Clark, 1947, p. 169). One of their last papers on the subject revealed an interesting fact: *Southern* — and so *more* segregated — children showed less of a tendency than Northern children to prefer the white doll over the brown one (Clark & Clark, 1947, pp. 177–178; but see also Cross, 1991). As Brooks (1996) notes, this "tends to indicate that racial isolation (even when tied to *de jure* segregation) provides a better environment than does racial integration for the development of a positive African American identity" (p. 21).<sup>20</sup>

To be sure, I do not claim that blacks will necessarily have low self-esteem in an integrated society. To see one important reason why, consider the fact that a person may have high self-esteem in one domain but low self-esteem in another: high self-esteem in athletics, for example, but low self-esteem in academics. Because this is true, having low self-esteem in one domain far from guarantees that a person will have low global esteem, for even if her low self-esteem reflects her poor appraisal in that domain, her self-esteem in other domains may be more important to her global self-

<sup>&</sup>lt;sup>20</sup> See also (Twenge & Crocker, 2002), which found the African American self-esteem advantage to be larger in the Southern US (p. 380). As Twenge and Crocker go on to note, what they call "racial density" (p. 387) is associated with increased self-esteem for other minority groups as well. Asian Americans, for example, generally have lower self-esteem than white Americans, but their disadvantage is less in the Western US, where Asians make up a higher percentage of the population (p. 382). Thus they suggest that "for minority groups, there may be a benefit to having frequent contact with people of your own race or feeling that you are included in a critical mass of people of your own race" (pp. 387–388).

esteem. But notice that a domain assigned a low value may be lowly valued precisely because of the person's poor appraisal in it; thus if someone is poorly appraised as a student, but highly appraised as an athlete, she may come to value athletics more than she values academics (Crocker et al., 1998, pp. 528–530; Harter, 1993, pp. 101–109). This suggests that even if blacks are devalued in the domain of physical attractiveness, they may come to place more value on other domains, which will then become more important to their self-esteem.<sup>21</sup>

Now though I have claimed that African Americans have had some success in protecting themselves against stigmatization and phenotypic devaluation, I do not want to claim that they have been wholly successful in this regard. In particular, phenotypic devaluation — as expressed in any context or media, not just interpersonal relations — seems to have already had a great impact on them, influencing their judgements of physical attractiveness, especially for women, who are more likely to be judged on their physical appearance (see, e.g., Neal & Wilson, 1989; Parmer et al., 2004; Russell et al., 2013). As one team of researchers has bluntly stated, "females with lighter skin, long silky hair, and White European facial features are considered as the standard for physical attractiveness among African Americans" (Parmer et al., 2004, p. 234). We can perhaps see the clearest evidence of this in the tendency of many African American women to straighten their hair.

My claim, then, is that racial integration threatens black self-esteem because it dissolves what we have reason to believe has been protective of African American self-esteem: group boundaries. It is the strong sense of themselves as members of a distinct group, which they positively evaluate, that seems to allow African Americans to have high self-esteem in the face of stigmatization and phenotypic devaluation. Moreover, since high self-esteem is intrinsically valuable, and studies have shown it to predict important life outcomes, concern for self-esteem provides one reason for blacks to resist integration (Orth & Robbins, 2014).

This argument, to be sure, assumes that there will be a *need* to protect black self-esteem even in an integrated society. After all, if there is no or little stigmatization or phenotypic devaluation in an integrated society, then it hardly matters if blacks will not be in a position to employ the psychological strategies that have been protective of their self-esteem in a segregated one. Some advocates of integration have in fact defended it, in part, because of its claimed prejudice-reducing effects (e.g., Anderson, 2010). Below I will argue that while integration may neutralize the threat to black self-worth posed by stigmatization, there is more reason to doubt whether it could neutralize the threat to it posed by phenotypic devaluation.

## III.

I have argued that integration threatens black self-esteem because it will dissolve the positive group identity that has been protective of it. In the next section, I explain three other ways that integration threatens black self-worth. Before turning to these

<sup>&</sup>lt;sup>21</sup> This may lead some to argue that, as long as there are a number of domains available to blacks to be esteemed, it should not matter if they are generally not esteemed for their physical appearance (Cf. Fourie, 2015, pp. 99–101). Needless to say, I reject this suggestion.

arguments, however, there is an issue that needs to be discussed first. It will be noted that the above argument assumes that integration will somehow involve assimilation, but that so far I have said nothing to defend this assumption. I do so in this section.

Before I begin, two points of clarification are in order. First, while my argument does identify some negative consequences of assimilation, it does not claim that assimilation — even black assimilation — per se is objectionable. Others have made such claims. For example, some critics of integration argue that it 'deprives' blacks of culture, and through this, of self-worth (McGary, 1999). <sup>22</sup> Other critics demand that blacks be respected as equals "from a distinct and valued subculture," and reject integration as incompatible with this demand (Adams, 2006, p. 294). <sup>23</sup> My argument is different. Though it places great importance on a distinct group identity, it does not claim that this distinct identity is valuable for its own sake. <sup>24</sup>

Second, we should be clear about what exactly the dispute regarding integration and assimilation is about. One reason integration has fallen out of favour among many African Americans is precisely its association with assimilation. As Adams (2006) explains, "Customarily, integration is seen as synonymous with assimilation, and successful integration necessarily entails the obliteration of a unique black identity" (p. 276). In light of this, recent defenders of it have been at pains to deny that their vision of it involves assimilation, emphasizing that it does not *require* it (Adams, 2006; Anderson, 2010; Stanley, 2017). For example, Anderson (2010) has stressed that, on her account, integration should not be conflated with a colourblind ideal that would "abolish" racial identities (p. 114). On her view, whether distinct racial identities would remain would be left to individual choice. This is all well and good, but the most pressing question regarding assimilation is not whether any particular vision of integration would require it, but whether *in practice* it would lead to it.<sup>25</sup> Thus my discussion focuses on this question.

Integration, as I have been understanding it, is the joining together of the members of distinct groups into some form of enduring association despite their differing group membership. It can be more or less comprehensive in scope, and can take place in different domains. I begin by listing some of these. First, *civic integration* includes both legal residency and "a disposition to engage with fellow citizens on matters of mutual concern" (Merry, 2013, p. 8). *School integration* is integration in schools, both public and private. *Economic integration* is the integration of all sectors

<sup>&</sup>lt;sup>22</sup> McGary (1999) does not endorse this argument. Kwame Ture and Charles Hamilton (1992) make a version of it: "The racial and cultural personality of the black community must be preserved and that community must win its freedom while preserving its cultural integrity ... No person can be healthy, complete and mature if he must deny a part of himself; this is what integration has required thus far" (p. 55). See also (Stubblefield, 2005, Ch. 3, esp. pp. 100–102).

<sup>&</sup>lt;sup>23</sup> Adams (2006) does not endorse this argument. Other arguments are discussed in (Shelby, 2005, Ch. 5). Although my argument has some affinity with the fourth tenet of what Shelby calls "black cultural nationalism" (p. 165), as a whole, there are more differences than similarities between this view and my own

<sup>&</sup>lt;sup>24</sup> There are two distinct claims about black identity that should be distinguished: i) that such an identity is valuable as an end in itself, and ii) that such an identity is valuable as a means of fostering a positive identity capable of countering certain threats to it. (Adams (2006, pp. 296–297) makes a similar distinction.) I am here concerned to defend ii), not i).

<sup>&</sup>lt;sup>25</sup> Anderson (2010, p. 115) was not unaware of this other question regarding integration.

of an economy so that, in a maximally integrated economy, there would be no sector associated with some racial groups and not others. *Residential integration* is integration in residential neighbourhoods. Finally, *psychological integration* is the psychological orientation, on the part of members of some group, of feeling a part of some other group.<sup>26</sup> It essentially involves the dissolution of the group boundaries that I have argued are so important to black self-esteem. In fact, it is the domain most crucial, in my view, for protecting black self-worth more generally.

I claim that assimilation in conditions of phenotypic devaluation or stigmatization makes it likely that blacks will internalize their devaluation. I understand 'assimilation' to mean the process, or end result of the process, whereby a person or group of persons loses the distinctive behaviour, outlook, and values of one group and gains those of another. Now integration in different domains will exert assimilative pressures of differing strengths because of differences in the closeness of the interracial contact that they involve. Economic integration, for example, involves closer sustained contact than civic integration, and so exerts stronger assimilative pressures. Assimilative pressures will also be greater when the group is integrated in multiple domains at once since this means more sustained close contact. In my view, given the closeness of the contact that it involves, residential integration — especially when combined with integration in other domains, like school integration — significantly increases black vulnerability, and it does so because it increases the risk of psychological integration.

Key to my argument is the fact that we are dealing with members of a group who are (as we are assuming) a small numerical minority. When the members of a small numerical minority integrate with a numerically larger group, they will most commonly be in contact with members of the other group. This is important because people's cultural orientation is most influenced by those with whom they have had the most sustained close contact. This means that if blacks are devalued by a non-black numerical majority, sustained close contact with them will mean assimilating into their culture, and possibly internalizing the group's devalued status.

Now it may be wondered, if residential integration significantly increases black vulnerability, why may we not limit integration to domains that involve less close contact? This raises a point that some of the new integrationists have stressed, i.e., that integration does not require perfect mixture in all spaces and institutions (Anderson, 2010, pp. 113–114; Stanley, 2017). So some allow that racial clustering in residential neighbourhoods, for example, may continue to exist.

The problem with this is that it raises questions about whether an integration that is limited in this way can provide the benefits that integrationists have argued it would bring. For example, one of the most important selling points of integration is that it will reduce black stigmatization. But it may be that nothing less than full residential integration, for example, can have this effect. (I discuss this issue more in Section V.) However, even such halfway measures may be enough to effect significant assimilation. Although residential integration increases the risk of psychological integration more than any other (single) domain, it is not the only route to this result. In short, in-between forms of integration jeopardize the benefits that integration is

 $<sup>^{26}</sup>$  I first encountered the concept of "psychological integration" in (Merry, 2013, p. 8, emphasis omitted).

supposed to bring, but without ensuring that blacks would be spared the harms of assimilation.

Notice that I do not claim that residential integration (even when combined with school integration) will *inevitably* lead to assimilation. For one thing, blacks may simply want to maintain a distinct identity, and this desire can help serve to limit its assimilative effects. My argument doesn't deny this. I do not claim that a racial group, however unwilling it may be to fully integrate, will still inevitably lose its distinct identity. Instead, and as I noted in the introduction, I am dealing with the special case of a group that is willing to integrate.

Another possibility is that continuing discrimination may lead blacks to adopt practices that have the effect of limiting their assimilation. Adams (2006) at one point appeals to just this possibility: "individuals might identify as black even outside predominantly black communities because of an appreciation for past discrimination and the ongoing challenges that blacks face" (p. 299). But this is actually troubling since it suggests that blacks might still be stigmatized in an integrated society. If integration is to be viable at all, it must be compatible with black self-worth, and if there is any hope for it to be compatible with black self-worth, it must be true that blacks would generally — if not immediately, then eventually — not be stigmatized in an integrated society. So if this is what limits the assimilative effects of integration, it is reason to reject integration itself. We should also keep in mind the possibility that its assimilative effects may at some point become irreversible, so that efforts to hold on to a distinct identity will be unsuccessful, a little too late.

A common strategy of integrationists is to contrast assimilation with what they call 'mutual transformation,' and to claim that their ideal of integration involves the latter, not the former (Adams, 2006, pp. 302–304; Stanley, 2017, Chs. 2 and 3). The idea is that instead of blacks simply assimilating into white spaces and institutions, they can "contribute meaningfully to the shape and content of integrated spaces and institutions," so that integration would "defy assimilation and reflect the aspirations of both blacks and whites" (Stanley, 2017, p. 74). But if blacks are a numerical minority, it is hard to see why even reconstituted or transformed spaces and institutions would not still mostly reflect the aspirations of non-blacks. And even if they didn't, it is hard, for the same reason, to see why they would avoid phenotypic devaluation, at least, with all of its implications for self-worth. Blacks may generally still be relationally devalued because of their phenotypes.<sup>27</sup>

#### IV.1

I have argued that, given the assimilative pressures that it will exert, integration may undermine black self-esteem. I will now argue that it may compromise black self-respect. Self-respect and self-esteem are clearly distinct. When a person is accused

<sup>&</sup>lt;sup>27</sup> I am not claiming that judgements of attractiveness inevitably have a same-race bias, and that since blacks are a minority, they will always be judged less attractive. Rather, my point is that the members of a numerical minority group who are generally judged to be unattractive by the members of a numerical majority will not suddenly be judged differently if they have a role to play in reconstituting certain spaces and institutions that members of that same numerical majority will also have a role in reconstituting.

of lacking self-respect, the accusation is *not* necessarily that he lacks self-esteem (Thomas, 1995, pp. 256–257). In addition, a person can have poor self-esteem while having a robust self-respect, and vice versa. For example, a person may feel bad about himself, but still adamantly refuse to do things that he considers beneath him. Both self-respect and self-esteem are forms of self-worth, but as Laurence Thomas (1995) says, self-respect is "a more fundamental sense of worth than self-esteem" (p. 261; see also Dillon, 2013; Sachs, 1981). The following self-respect-based argument thus provides a stronger reason to oppose integration than the self-esteem argument.

For many, the kind of worth that self-respect involves has to do with one's moral status or standing as a person, in particular with the idea that one's moral status is equal to that of others, so that one is no less entitled to basic rights (Boxill, 1992, Ch. 9; Hill, 1995a; Thomas, 1978, 1995). So for Thomas (1978), for example, self-respect "consists in having the belief that one has and is worthy of full moral status. And to believe this is to believe that one is as worthy as any other person of the set of moral rights to the recognition of which persons are entitled" (p. 309).<sup>28</sup> And since, in his view, this belief can only be sustained when the social institutions of one's society affirm this status, self-respect requires the equal distribution of rights and liberties (Thomas, 1978; see also Hill, 1995a; but see also Hill, 1995b, where he retreats from a purely rights-based account of self-respect). In my view, however, while it is certainly true that self-respect involves recognition of one's equal moral status, there is more to it than that.

Self-respect is a robust sense of one's value or worth (moral and otherwise) that indisposes one to act in certain ways or tolerate certain forms of treatment. The self-respecting person regards the relevant treatment and acts to be in some sense beneath her, and where feasible refuses to perform or tolerate them. Of course, there may be circumstances when a person can permissibly (i.e., consistent with self-respect) tolerate objectionable treatment or act in ways normally inconsistent with self-respect; cases of threats to life (to take the most extreme case) are presumably one such example. These actions and inactions do not imply an insufficient appreciation of worth, or insufficient steadfastness. However, a person can't act in certain ways or tolerate certain forms of treatment in order to avoid some minor setback to well-being (Hill, 1995a, pp. 84–85). These points suggest that our duty of self-respect should be understood as at least moderately demanding, and possibly very demanding, but that there is surely a point at which we can make it too demanding.

Self-respect is a *robust* recognition and appreciation of one's worth, and what follows from this with regard to how one acts and what one tolerates. Even when one duly appreciates one's worth, the appreciation may not be sufficiently robust, and if this is the case, the person will fail to be self-respecting. As Christian Schemmel (2019) has recently emphasized, a person's self-respect must be robust, for "being able to deal with adversity is a constitutive part of having self-respect" (p. 633). This means that the self-respecting person doesn't crumble in the face of adversity.

<sup>&</sup>lt;sup>28</sup> This is a Kantian view of self-respect. For discussion, see, e.g., (Holberg, 2017; Massey, 1983; Stark, 1997).

She does not buckle at the first appearance of unfavourable circumstances. Rather, she remains *steadfast*.

Now it may be wondered why a person's self-respect, as opposed to her self-esteem, should be affected by how she is treated by others (Cf. Margalit, 1996, pp. 24–25). Howard McGary (1999) has written:

A crucial aspect of self-esteem is the fact that the opinions of others are crucial to a person's assessment of whether or not he feels that he is worthy of esteem. Self-respect, on the other hand, does not necessarily depend upon the assessment of others. In fact, the person who stands steadfast in spite of the negative assessment of others might be thought, under appropriate conditions, to epitomize the self-respecting person. (p. 49)

McGary is surely right that self-respect does not necessarily depend on the assessment of others, and that it is the person who stands steadfast in the face of negative assessments who epitomizes the self-respecting person. Self-respect, as I have said, is robust. But one can of course affirm that self-respect should be at least somewhat resistant to the negative assessments of others, while also affirming that it can be negatively impacted by such assessments, at least some subset of them.

At this point, however, we must consider a more uncompromising view of the demands of self-respect. It can be *denied* that negative assessments can rationally impact self-respect, for it is possible to claim, as Colin Bird (2010) has claimed, that however often people typically let the assessment of others undermine their self-respect, in doing so they act *irrationally*, for "what others do cannot constitute a *good reason* to lose one's self-respect; it is always possible to resist outside assaults on one's worth" (p. 19; emphasis added). This uncompromising stoical view regards threats to a person's self-respect — no matter the circumstances that constitute them — as always in principle capable of being resisted, and so any failure of resistance as irrational. To be sure, those who endorse this kind of view need not indiscriminately condemn all those who let the assessments of others compromise their self-respect; they might accept that in some circumstances we may not properly blame those who succumb to negative assessments (Bird, 2010, p. 19).

This position is flawed. To see this, note that there are surely at least three kinds of positions that one might take on the relation of a person's self-respect to the assessments (or treatment) of others: first, one might claim, like Bird, that it is *always* the agent's responsibility to maintain her self-respect in the face of others' negative assessment, no matter how poor, widely shared, or rudely expressed; second, one might claim that *any* negative assessment, by however few people, in whatever circumstances, and however expressed, can provide a good reason for a person to lose her self-respect; and finally, one might take a more middle ground position, which holds that it is neither always the agent's responsibility to maintain her self-respect nor should it be so vulnerable that even a single person politely conveying a mildly negative assessment can undermine it.

I take it that the first position is too demanding, the second is insufficiently demanding, and that the correct view is some version of the third. But Bird (2010) strangely overlooks it. The only alternative to allowing any negative assessment

undermine a person's self-respect, he appears to think, is the stoic one. This is why he can claim that it is an advantage of that alternative that "it doesn't hold self-respect hostage to the utopian idea that these forms of abuse will easily disappear from the earth" (p. 38). Of course, there are a variety of positions that one might take consistent with making self-respect neither completely vulnerable nor completely invulnerable; within these broad constraints, we can make self-respect more or less demanding. I hold that self-respect is a duty that makes relatively strong demands on individuals; we should not let just any negative assessment undermine our recognition or appreciation of our worth. But negative assessments can surely be sufficiently poor and widespread that they can reasonably undermine our self-respect. That is, they can be such that a person whose self-respect is undermined by them does not thereby show himself to be irrational or insufficiently steadfast.

The conception of self-respect that I have described is one that has to do with what Stephen Darwall (1977) has called "recognition self-respect" (p. 38). This is a form of what Robin Dillon (2018) calls "status worth," which she defines as "worth that derives from such things as one's essential nature as a person, membership in a certain class, group, or people, social role, or place in a social hierarchy." As she notes, most discussions of self-respect focus on a person-based status. I assume that all persons, as persons, should recognize and appreciate themselves as having worth, but I believe that in thinking about self-respect it is helpful to think beyond general personhood and consider how persons, as distinct, concrete individuals, live this status.

The reason for this is that we are not simply persons but persons with distinct group identities: we have racial, sexual, and other identities. This is to say that though we all have status worth as persons, we also have status worth as persons who are black, as persons who are women, as persons who are gay/lesbian, and so on. To be clear, I am not claiming that this identity-based status worth is a worth over and beyond our worth as persons; it is a worth that we have simply in virtue of being persons. So it trivially follows from our status worth as persons, who have various group identities. I highlight it because some persons face threats to their self-respect because of their identities; for example, some threats to person-worth are distinctive because a person is a woman, or is gay. Thus if many feminists are right that a woman who is unduly deferential to her husband lacks self-respect, this is arguably not simply because a married person shouldn't be unduly deferential to his or her spouse, but rather is closely connected to the fact that she is a woman (Dillon, 1992; Hill, 1995a; Holberg, 2017; Stark, 1997). In a similar way, what is non-self-respecting about the behaviour of the figure of 'Uncle Tom' is arguably closely connected to the fact that he is black (Dillon, 1992; Margalit, 1996, pp. 35-39; Stark, 1997; Thomas, 1978, 1995). So though I am not claiming that a person's race, sex, sexual orientation, and so on ever provides what we may call a basic ground of self-respect, a person's self-respect may be threatened in certain ways because of her identity. Moreover, when a person is able to continue living in light of a robust appreciation of her worth in the face of these distinctive identity-based threats, it seems fitting to speak of that person as having racial self-respect, sexual self-respect, and so on. My claim, then, is that racial integration threatens blacks' racial self-respect.

Before proceeding, however, let me say something about how self-respectinconsistent actions relate to the state of lacking self-respect. There are two kinds of relations. One, what I call the 'constitutive' relation, holds that some acts are such that they constitute the person performing them as lacking self-respect; if someone tolerates a certain act of abuse in order to achieve some minor good, for example, we may view this not merely as evidence that she fails to appreciate her worth, and so evidence that she lacks self-respect, but as constituting her, at least at that time, as non-self-respecting. On the other, which I call the 'evidential' relation, the relevant acting is merely inconclusive evidence that the person lacks self-respect. This evidence might be inconclusive because i) it is possible that it does not in fact flow from an appropriate (self-respect-compromising) source (i.e., a lack of steadfastness), and ii) self-respect is usually character-trait-like in that it is relatively stable over time, but just as a person with, say, the character trait of honesty may occasionally act dishonestly, so a self-respecting person may occasionally, at least for some actions, act in a self-respect inconsistent fashion.<sup>29</sup> Self-respect-inconsistent actions can relate to the state of lacking self-respect in both ways: some acts count as evidence that the person performing them is non-self-respecting, and some acts constitute the person as non-self-respecting.

Implicit in my account thus far is that self-respect-inconsistent-seeming actions are of two kinds: they can be the result of a failure to *recognize* or *appreciate* one's worth, and they can be the result of a failure of *steadfastness*. Self-respect-inconsistent acts that are failures of steadfastness always constitute the person as non-self-respecting. If one fails to be sufficiently steadfast in the face of threats to self-respect, one has by definition failed to be sufficiently resolute, and so acts that flow from insufficient steadfastness always constitute one as non-self-respecting. By contrast, self-respect-inconsistent-seeming actions that are failures of recognition or appreciation of worth can either constitute the person as non-self-respecting or be evidence that he lacks self-respect.

Let us turn, then, to how integration can compromise black self-respect. Now when individuals sense that they are relationally devalued, this will likely motivate efforts to enhance their relational value (Cf. Leary & Allen, 2011; Leary & Baumeister, 2000, pp. 38–39). Actions taken in this regard raise questions of self-respect. For example, being devalued because of one's race may lead a person to act in a servile manner, as she recognizes the devalued status of her group, and attempts to ingratiate herself in the out-group by acting in this way. Such a person, it is worth adding, may sincerely reject any suggestion that she is not of equal moral status, and so not equally deserving of moral rights. This suggests that lacking self-respect need not involve the belief that one is not of equal moral status (Cf. Hill, 1995b).

So servilely ingratiating oneself with an out-group is one way that one can act in a self-respect-inconsistent manner, and so may be non-self-respecting. Another is trying to *pass as* a member of an out-group — that is, hiding one's membership in a group. A person can pass with regard to race, ethnicity, religion, sexual orientation, and more, and the passing may be temporary or permanent. With regard to racial passing, in the nineteenth century and the first half of the twentieth century, many

<sup>&</sup>lt;sup>29</sup> Cf. Dillon (2013): "to respect oneself is not merely to have a certain attitude but to be a certain kind of person; to lack self-respect is to be another kind of person" (p. 4776).

light-skinned African Americans passed as white as a way of escaping first slavery, and later the stigma and restrictions of the American caste system (Hobbs, 2014; Kennedy, 2003, Chs. 7 and 8).

Now given my assumptions about the black population whose integration we are considering, it is clear that passing for white is not an option for them. But there is a kind of passing that even phenotypically stereotypical blacks may be tempted to try. Suppose, for example, that knowledge of racial self-identification by *itself* — that is, quite apart from a person's phenotypic traits — affects how individuals are treated. In that case, some stereotypical blacks may be tempted to try to pass as members of a racial category defined in part by lesser stereotypicality (e.g., mixed race). Something like this seems to happen in Brazil, where sociologist Elizabeth Hordge-Freeman (2015) finds that black Brazilians tend to "whiten themselves by using intermediate color terms" (p. 205). As one of her informants says, "We call ourselves *morenos* so that we do not have to say *negro*" (pp. 146–147; see also Twine, 1998, Ch. 6). We may want to say that in doing so such blacks constitute themselves as non-self-respecting. Although blacks remain highly stigmatized, we might demand that they be more steadfast than that.<sup>31</sup>

At this point, we should consider the question of self-respect and blame. Should we blame people for not having self-respect? Not always. Here we return to the distinction between self-respect-inconsistent actions that are failures of recognition or appreciation of worth, and those that are failures of steadfastness. I believe that failures of steadfastness are always blameworthy, but that failures of recognition/appreciation are not. Circumstances can be such as to make it very difficult to recognize and appreciate one's worth. If one is subject to widely shared and frequently expressed negative assessments, for example, this can make it difficult to develop and/or sustain an appreciation of worth, and so make it difficult to maintain one's self-respect. Blame in such cases seems inappropriate. I think that this can be the case in circumstances of widespread devaluation.

The discussion so far offers general lessons about the potential dangers of integration for blacks. But it also suggests a corresponding ideal that societies should strive to achieve: if a society's 'racial arrangements' make blacks want to be something other than black, then they are inconsistent with black self-worth and so should be unacceptable to them.<sup>32</sup> We should think of this as a particular application of a general race-

<sup>&</sup>lt;sup>30</sup> This is not as farfetched as it sounds. A study by Robert Reece (2016) found that a mixed-raced identity increased perceived attractiveness independently of phenotypic traits like skin colour. (This study, however, admittedly could not control for phenotypic characteristics like nose width.) Note that the analysis that follows is intended to also apply to blacks who alter their features, including hair texture, to appear to belong to an intermediate racial category.

<sup>&</sup>lt;sup>31</sup> I do not claim, however, that passing is always inconsistent with black self-respect. Whether an act is inconsistent with self-respect depends on both the nature of the act and the good or evil that it is intended to promote or avoid. Some acts are by their nature in tension with self-respect, but the evil that they are intended to avoid can be such that, all self-respect-related factors considered, they are not inconsistent with it. (This does not turn self-respect into merely one constituent component of the consequentialist good that competes with other components.)

<sup>&</sup>lt;sup>32</sup> I say that blacks should not *want to be something other than black*, not that they shouldn't want to be something other than black *for certain reasons*. Such a claim would imply that there are permissible reasons for not wanting to be black, impermissible reasons for the same, and some way of drawing the distinction. I

neutral test that societies must pass in order to be acceptable to the members of the various racial groups: these members should not want to be some other race. So my claim is that, in conditions of stigmatization or phenotypic devaluation, there is reason to doubt whether integration can pass what we may call the *same-race test*.

Another kind of act that is inconsistent with black self-respect concerns dating and marriage. For many, dating and marrying across racial lines are signs of admirable open-mindedness (Cashin, 2017). Yet I will argue that they may also be signs of the internalization of one's devaluation (Cf. Fanon, 1967, Ch. 2). Before proceeding, however, two points should be kept in mind. First, the meaning of instances of interracial dating and marriage may be different depending on the racial identity of the party being considered, especially if there are group status differences between the groups to which the parties belong. Second, people date and marry interracially for a variety of reasons, and not all of these raise questions of self-worth.<sup>33</sup> Just as it seems that an act that is permissible if done for one reason can be impermissible if done for another, so an act that is self-respect-consistent if done from one motive may be self-respect-inconsistent if done from another. Thus the interracial relationship of a black person with no self-worth issues should not be judged the same way as one in which the black person has such issues.

Now if you are a member of a devalued group, and you want to increase your relational value by disassociating oneself from the group, one way of doing so is to find some way of signalling to out-group members that, despite one's group identity, one in fact has high relational value. Dating and marrying interracially are ways of doing this. I claim that if one dates or marries interracially for this reason, one constitutes oneself as non-self-respecting.<sup>34</sup>

### IV.2

A third self-worth harm of relational devaluation is that it can lead to feelings of inferiority. Some philosophers have noted that one reason for objecting to inequalities of various kinds (status, esteem, etc.) is that they can generate feelings of inferiority. For example, T. M Scanlon (2003) has claimed that, among other reasons, we may object to inequality because "it is an evil for people to be treated as inferior, or made to feel inferior" (p. 204; see also Fourie, 2015; Scanlon, 2018, Ch. 3). This is clearly because it is intrinsically bad to have such feelings. I claim that being a member of a devalued group in an integrated society may generate them.

Inequality of phenotypic esteem may also be objectionable if it gives rise to *relations* of inferiority and superiority, even if this is not accompanied by feelings of

doubt that, in the domain of race, any such distinction can be made. I won't pursue this issue here, however. The point of discussing this ideal is to show its inadequacy for the reasons that I go on to discuss.

<sup>&</sup>lt;sup>33</sup> Consider some of the possible reasons why some persons might end up in an interracial relationship: because there are few same-race persons in the society; because they are strongly interested in an activity that few members of their racial group are interested in (e.g., blacks who are obsessed with medieval-style games); because they have characteristics that make it very hard to find compatible mates of any race; because they simply did not let race affect their choice of partners; and so on.

<sup>&</sup>lt;sup>34</sup> Again, the example of Brazil is instructive. Among black Brazilians there is a widespread desire for white, or at least lighter-skinned, partners (see Hordge-Freeman, 2015, esp. pp. 56–67; Twine, 1998).

inferiority. Above, in discussing Fourie's work, I noted that a society characterized by inequality of phenotype-based esteem is one that relational egalitarians may have reason to object to. I explore this further now. Relational egalitarianism is best explained by contrasting it with distributive egalitarianism, which was traditionally the dominant approach to equality. According to the distributive view, justice requires the equal distribution of something (e.g., resources or welfare), and the task for the egalitarian theorist is to figure out what this is. Like the distributive view, the relational view accepts that there is an ideal of equality that is required by justice, but understands it as one that in the first instance applies to certain kinds of interpersonal relationships among the members of society. As Samuel Scheffler (2015) puts it, it requires the establishment of a "society of equals" in which its members "relate to one another on a footing of equality" (p. 21). This kind of equality may have distributive implications, but they are indirect, flowing from a broader ideal.

There is much to be said in favour of relational egalitarianism. However, accounts of it often leave it unclear why individuals should want to avoid the unequal relations that it condemns. These accounts seem to suggest that these reasons bottom out in considerations of justice: relating on a footing of equality is good because it is what justice requires. But I believe that there are reasons beyond justice for establishing a society of equals, and, moreover, that individuals have reasons beyond justice for wanting to avoid unequal relations. I have in mind reasons of self-worth. We all have an interest in achieving and maintaining a sense of our own worth. But unequal relations may make this impossible. So, from this point of view, the main problem with unequal relations, at bottom, is that the parties in the unequal position may come to *accept* the inferior place that they are assigned in the relationship.

To see that integration in conditions of stigmatization and phenotypic devaluation can lead to objectionably unequal relations for this reason, consider the following passage from Hordge-Freeman's (2015) study of Brazil:

As I sit in on an art course for children, I observe Afro-Brazilian girls stroking the straight hair of their white and straight-haired peers during instructional time, during classroom breaks, and during lunch. Even as they participate in conversations with peers and teachers, they continue running their fingers through the straight hair of their white classmates.

During breaks, these young Afro-Brazilian girls run to the bathroom and douse their hair with water in order to achieve a look that is wet and wavy, rather than dry or "bad," and then they return to class. Their straight-haired peers, the objects of their attention, do not respond to the attention, and everyone acts as though this is a normal part of their daily activities. (pp. 91–92)

Part of what makes the scene depicted so tragic is that the Afro-Brazilian girls evidently accept the inferiority attributed to them by Brazilian society: whites and nearwhites have 'good' phenotypic traits, but they do not. Note, moreover, that they may

<sup>&</sup>lt;sup>35</sup> For example, Scheffler (2015) argues that we should avoid unequal relations because this is inconsistent with a society of equals, and we should have a society of equals because this is what justice requires.

accept this inferiority without *feeling* inferior. Indeed, one characteristic feature of accepting one's inferiority is that one is often blind to this fact.

#### IV.3

Importantly, the self-worth harms of relational devaluation can be brought about even when the devaluation is rooted in phenotypic devaluation, not just stigmatization. This is perhaps clearer with regard to self-esteem and feelings and relations of inferiority: a person's self-esteem can be negatively impacted by phenotypic devaluation, and the same can also give rise to a feeling of or acceptance of inferiority. Phenotypic devaluation can also lead to acts that suggest that the person is, or that constitute the person as, non-self-respecting. It can lead persons to act in servile ways. It can also lead to passing. Entering into an interracial relationship may also constitute a person as non-respecting when done because of phenotypic devaluation. A person might date or marry interracially not to raise her relational value, but that of her children. She may desire to reposition her family along what Hordge-Freeman (2015) calls the "phenotypic continuum" by having phenotypically less stereotypical (e.g., lighter-skinned) descendants — something, it is worth nothing, that can also be pursued endogamously with less stereotypical members of the group.<sup>36</sup> (Thus the same reasons that we have for objecting to some interracial unions, when entered into for certain reasons, also apply to some intraracial unions, when entered into for the same reasons.) This act is in serious tension with black self-respect, and indeed is among the most serious ways that a black person may manifest a lack of it. 37 Acting servilely and hiding one's identity are both ways that a black person might fail to live in light of a robust appreciation of worth, but acting on a desire — whether for oneself or one's children, or both — to be some other race, or some reduced version of one's race, is perhaps the quintessential way of doing this.<sup>38</sup>

Consider also this. Above, I noted that one important way of testing whether integration is consistent with black self-worth is to apply what I called the 'same-race test'; in this case, if a significant number of blacks would not want to be black in an integrated society, integration should be considered inconsistent with black self-worth, and so rejected. Consider now that phenotypic devaluation can make blacks not want to be black. If one is devalued because of one's phenotypic traits, a natural reaction is to wish that one had some other traits. If this is right, then the self-worth

<sup>&</sup>lt;sup>36</sup> The term "phenotypic continuum" is used several times in (Hordge-Freeman, 2015, e.g., pp. 48, 102). For clear evidence that many black Brazilians make choices about dating and marriage with a view to repositioning their families along the phenotypic continuum, see (pp. 56–67). The practice is also widespread in the US, the West Indies and other places in Latin America. For the US, see (Russell et al., 2013, Ch. 6); for the West Indies, see (Lowenthal, 1972, esp. pp. 97–98, 137–138, and 257–258); and for Columbia, see (Wade, 1993, esp. Chs. 4 and 16).

<sup>&</sup>lt;sup>37</sup> It may be claimed that since what is insufficiently appreciated are certain features, not race itself, this can't be a case of a lack of *racial* self-respect. But, if I grant this distinction, I can still maintain that it indicates a lack of a phenotype-based self-respect.

<sup>&</sup>lt;sup>38</sup> One may desire to have less stereotypical offspring both for their sake and for one's own: for them to escape the worst of stigmatization and/or phenotypic devaluation, and for oneself to be esteemed for positioning them to do so. Cf. (Russell et al., 2013): "A program director at an urban clinic told us that she has witnessed teenage Black girls who intentionally got pregnant by a light-skinned boy in the hope that a light-skinned baby would bring them love — both from their mothers as well as their babies" (p. 140).

harms of relational devaluation may be rooted in phenotypic devaluation, and not just stigmatization.

However, we should recognize that even when phenotypic devaluation makes persons want to have different phenotypic traits, it may not always make them want to be a different *race*. If one is devalued because of one's phenotypic traits, one may wish to have different *traits*, but this need not mean that one wishes to be of a different race. This may be the case, but it may not. In her study of Brazil, many of Hordge-Freeman's (2015) interview subjects clearly seem to wish that they were closer to the white end of the phenotypic continuum, but some expressly deny that they want to be white; as one nine-year-old girl says, "If I could make any wish, I would wish for long blonde hair and blue eyes. But I don't really want to be white" (p. 119).<sup>39</sup>

But even if phenotypic devaluation makes blacks merely want to be less stereotypical phenotypically, rather than being some other race, this too is objectionable. <sup>40</sup> It cannot be that an arrangement of society is unacceptable if it would make persons wish to be some other race, but is acceptable if it makes them want to be less stereotypical members of their own race — in our present case, unacceptable if it would make blacks not want to be black, but acceptable if they want to be less black. <sup>41</sup> The intuitions that tell against the one seem also to tell against the other. Further, the same-race test seems apt to focus attention on self-worth harms less likely to endure in an integrated society and to fail to notice those unlikely to fade away. For if blacks are no longer stigmatized in such a society, as integrationists must argue, then blackness as a *racial category* would not be something marked as deviant and blacks should not want to be something other than black. Indeed, blackness might be something that is viewed with considerable pride. But such a view of blackness is compatible with more stereotypical black phenotypic traits being regarded as misfortunes, if only for aesthetic reasons. <sup>42</sup>

All of this suggests that the same-race test should be supplemented with an additional test more particularly concerned with phenotypic traits. This additional test would require that for a society's racial arrangements to be acceptable to the members of the relevant racial groups, they should not want to have different phenotypic traits; in particular, they should not want to have less stereotypical traits. Let us call this the *phenotypic test*. My claim, then, is that in conditions of phenotypic devaluation, an integrated society won't pass the phenotypic test, even if it can pass the same-race test.

<sup>&</sup>lt;sup>39</sup> Hordge-Freeman (2015) reports that in Salvador, Bahia, "*Morenidade* is considered an attainable approximation of Western whiteness characterized by light-brown skin (rather than white skin) and soft brown hair that bounces" (p. 48).

<sup>&</sup>lt;sup>40</sup> Some authors wrongly make much of the fact many blacks seem to endorse a mixed-race beauty ideal, as opposed to a white one, as if this were not objectionable. See, e.g., (Gordon, 2019; Patterson, 1997, pp. 70–71).

<sup>&</sup>lt;sup>41</sup> I do not mean to suggest that blacks who are less stereotypical phenotypically are necessarily less black. But one way of being less stereotypical is being less black.

<sup>&</sup>lt;sup>42</sup> This is likely to be worse for stereotypical black women, who as women are more likely to be judged for their appearance, and who as stereotypical will likely be more harshly evaluated since stereotypicality seems to more negatively impact judgements of black female attractiveness (see, e.g., Burke et al., 2013; Lewis, 2011).

#### IV.4

Now the self-esteem argument developed in Section II and the feelings of inferiority argument of Section IV.2 work by providing individual blacks with prudential reasons to avoid racial integration — integration is counter to their good or well-being. The self-respect-based argument, however, proceeds differently (so too does the relations of inferiority argument). It holds that individual blacks have *moral* reasons to resist integration. The idea is that we have a moral duty to robustly appreciate our own worth, and to act in light of this appreciation (Cf. Hill, 1995a, pp. 89–91). Moreover, since we have a moral duty to maintain our self-respect, we have moral reasons to avoid that which tends to undermine self-respect. Hence blacks have moral — in addition to prudential — reasons to avoid integration.

To be sure, some deny that we can have duties to ourselves, and if they are right, we of course can't have a duty to respect ourselves (Singer, 1959). This is surely not the place to take up this issue. I therefore make my claim conditional: *if* we can have duties to ourselves — and duties of self-respect are surely plausible candidates for such duties, if they exist — then blacks may have duties to resist integration (Cf. Hill, 1995a). Further, since some acts may be morally valuable whether or not they are imposed by moral duties, so that we may have moral reasons to perform them even if these reasons fail to amount to duties, blacks may have moral reasons to resist integration even if these reasons do not amount to duties.

These arguments, it should be noted, do *not* question the benefits of integration for which integrationists like Anderson (2010) have argued; for all that I argue, integration may indeed be justice/democracy-enhancing, even if it is not quite the imperative that she has claimed. Rather, I hold that self-worth is so valuable, and consequently the potential self-worth harms of integration so significant, that they may outweigh the justice/democracy benefits it is said to bring. Self-worth is a great good. Self-esteem, as I have suggested, is intrinsically valuable, and it has been linked to important life outcomes; avoiding feelings of inferiority is an obvious and significant good; self-respect is an extremely important form of self-worth (Cf. Rawls, 1999, pp. 386–391), the absence of which precludes a truly flourishing life; and, as the precondition for the sense of agency needed to pursue so much of what is valuable in life, not accepting that one is anyone's inferior may be the most important form of self-worth of them all.<sup>44</sup>

And yet, as great as the good of self-worth is, and as bad as self-worth harms can be, self-worth does not have absolute overriding weight such that it cannot be outweighed by anything else. So it is possible for the self-worth harms of integration to be outweighed by its benefits. Suppose that if blacks integrate, a protracted and deadly civil war, in which millions — blacks and non-blacks alike — lose their

<sup>&</sup>lt;sup>43</sup> This is not inconsistent. Having a duty to resolutely respect ourselves does not preclude taking action to avoid being in situations that will predictably undermine our resoluteness. (In addition, the situations may undermine, not our resoluteness, but our appreciation of our worth.)

<sup>&</sup>lt;sup>44</sup> These claims about the value of self-worth are entirely race-neutral ones: self-worth is an important value for all racial groups. But I suspect that certain facts about a group's history may amplify the significance of some self-worth harms, giving the group additional reasons to resist them. (These additional reasons should be added to more ordinary reasons when weighing the harms of integration against its benefits.)

lives, is avoided. If so, the harms of integration are probably outweighed by its benefits. And this is so even if a worst-case scenario obtains as far as these harms are concerned. But such fantastic scenarios are unlikely to obtain in the world in which we live. In this world, the benefits are almost certain to be much more modest. In addition, the example has a special feature — great benefits to non-blacks — that is also unlikely to obtain. In this world, any benefits to others are likely to be negligible.

The benefits to blacks, however, may be great, as the example itself suggests. But again, I think that we are entitled to set aside such unrealistic scenarios. If we do that, it is very hard to see how the benefits of integration could be so great as to outweigh the harms, assuming something at least reasonably close to a worst-case scenario. So if self-worth is as valuable as I have argued that it is, even the fullest realization of the benefits that we can realistically expect could not outweigh the harms that we can reasonably expect. In comparison to *these* (realistic) benefits, it *does* have absolute overriding weight. An increase in an individual's economic standing, for example, could never outweigh the harm of a severely damaged sense of self-worth. 45

So I argue that blacks have prudential and moral reasons to resist integration, even if it is true that integration is justice/democracy-enhancing. Now it could be argued that this pitting of self-worth, and in particular self-respect, against justice wrongly assumes that a society can rightly be said to be just while some of its members — defined by a group identity such as race no less — lack a form of self-respect. This kind of objection appears most at home in a view that sees self-respect as exclusively threatened by social institutions, at least if it is combined with the view that the *only* subject of social justice is social institutions. <sup>46</sup> For if social institutions are the only subject of justice, it seems plausible to hold that they should be so arranged that they secure for individuals a good as important as self-respect. But my argument shows that this narrow view of the threats to self-respect is mistaken; various forms of interpersonal interaction, in addition to social institutions, can undermine it. Given this expanded view of the threats to self-respect, there is correspondingly less reason to think that justice is inconsistent with *all* actions that tend to undermine it. <sup>47</sup>

<sup>&</sup>lt;sup>45</sup> Now it could be argued that I should give some accounting here of the benefits that Anderson and others have claimed that integration would bring. (One anonymous referee strenuously argued for the need to do just this.) But I believe that my position obviates the need for a more detailed accounting. For if we restrict our view only to scenarios that are likely to obtain, then no matter what those benefits turn out to be — even if, say, Anderson's vision of racial equality and a more robust and responsive democracy were fully realized — they cannot outweigh its harms — if, that is, these are as significant as what I imagine. (Whether that is the case depends on what the facts are regarding stigmatization and phenotypic devaluation — facts that we cannot examine in any detail here.)

<sup>&</sup>lt;sup>46</sup> The idea that it is social institutions, and social institutions alone, that threaten self-respect might be the view of Thomas (1995), although he doesn't strictly imply it. On the "basic structure" as the subject of justice, see (Rawls, 1999, pp. 6–7).

<sup>&</sup>lt;sup>47</sup> Notice that the claim is not that considerations of social justice do not govern individual choice in the private sphere, but rather that, given the expanded view of the threats to self-respect, it is implausible to think that *all* actions that may have the effect of undermining individuals' self-respect are unjust.

#### V.1

I have argued that integration will intensify the self-worth harms of stigmatization and phenotypic devaluation, and that these harms may be so significant that they may outweigh the benefits that integration is often said to bring. But for all that has been said, my argument may seem to simply assume that an integrated society will not in certain ways be much different from a segregated one. After all, and as I noted in Section II, one of the central virtues of integration, according to its champions, is precisely its prejudice-reducing benefits. So this issue requires examination.

To its advocates, integration is not short of virtues. It will, they say, overcome unjust group inequality, end mutual mistrust, and more fully realize a democratic society (Adams, 2006; Anderson, 2010). But what interests us here are claims that it can improve interracial relations by reducing black devaluation. As Anderson (2010) says, "Integration should reduce antiblack prejudice, stigmatization, discrimination, and anxiety" (p. 123). Let us call these benefits 'prejudice-reduction benefits.' In arguing that integration has prejudice-reduction benefits, Anderson relies heavily on the theory of intergroup relations mentioned in the introduction: contact theory (pp. 123–127; see also Cashin, 2017, pp. 162–168). I will argue that there are limits to the benefits of contact.

Contact theory makes a simple claim: contact between members of different groups can reduce 'intergroup conflict,' especially if the following conditions obtain: i) the groups have equal status, ii) the groups meet in pursuit of common goals, iii) there is cooperation between groups, and iv) there is institutional support for contact. It is a general theory, and so the beneficial effects of contact are supposed to hold across groups defined by race, ethnicity, sexual orientation, age, and more. Moreover, the benefits are supposed to be generalizable with regard to out-group members not involved in the contact, other situations, and even other out-groups. In their meta-analysis of over 500 studies, Thomas Pettigrew and Linda Tropp (2011) examine all this, and much more. They find that intergroup contact does indeed have prejudice-reduction benefits that generalize broadly, and they convincingly defend the theory against many objections.

I want to concede the basic claim of contact theory, as applied to race relations: interracial contact can reduce prejudice. But this concession leaves us with two important questions. The first is whether prejudice reduction is a sufficient goal to aim for in the realm of race — and if it is not, whether contact can secure whatever would be sufficient. The second is whether intergroup contact has any unintended side effects. I will argue that prejudice reduction is not sufficient to address phenotypic devaluation, that interracial contact by itself seems incapable of addressing this, and that the most effective forms of interracial contact, as far as prejudice reduction is concerned, are just those that we have reason to believe will make blacks most vulnerable to the self-worth harms of phenotypic devaluation.

To substantiate these claims, more needs to be said about the theory. In particular, we need to ask exactly *how* intergroup contact is supposed to lead to prejudice

<sup>&</sup>lt;sup>48</sup> These conditions used to be taken as necessary, but work by Pettigrew and Tropp (2011) and others now suggests that, while they can amplify the benefits of contact, they aren't necessary for contact to produce benefits (Ch. 5).

reduction. Surmising that intergroup relations for groups meeting for the first time is liable to be marred by ignorance, anxiety, and lack of empathy, researchers have been primarily concerned with examining whether these three variables — knowledge, anxiety, and empathy — mediate contact and prejudice reduction. As Pettigrew and Tropp (2011) explain, "The straightforward idea is that successful intergroup contact will provide you with useful knowledge about the outgroup, reduce your anxiety in intergroup encounters, and help you to take the perspective of outgroup members and empathize with their concerns" (p. 77). Their work suggests that empathy and anxiety reduction are most important, and that knowledge is less so (Ch. 6). Their research also suggests that contact is more likely to reduce the affective components (dislikings, negative emotions) of prejudice than the cognitive components (beliefs and negative stereotypes) (Ch. 7).

Let us turn, then, to the first question, which is whether prejudice reduction is a sufficient goal. As I have been claiming, relational devaluation of blacks has at least two distinct sources: stigmatization and phenotypic devaluation. If its advocates are right, and interracial contact can reduce anti-black prejudice, stereotyping, and (race-based) discrimination, contact will be able to address relational devaluation rooted in stigmatization. But relational devaluation rooted in phenotypic devaluation *may* remain untouched. As I argued in Section I, there is no reason in principle why phenotypic devaluation might not remain after devaluation rooted in stigmatization is eliminated or significantly reduced. What's more, it is hard to see how interracial contact can even in principle reduce phenotypic devaluation through the familiar mediators of contact theory. It is hard to see how greater knowledge about blacks, more empathy with regard to them, or less anxiety in encounters with them can lessen it. Nor is it clear how other possible mediators could have this effect, or how contact *itself* could have it — our judgements of attractiveness do not seem to be a function of mere exposure or familiarity (see, e.g., Burke et al., 2013).

But there is one possibility that cannot be dismissed entirely. It has to do with the fact that mathematical averageness has been found to contribute to judgements of attractiveness (see, e.g., Rhodes, 2006). This raises the question: if average is attractive, what characteristics make a face average? The answer would seem to be the characteristics of the population of faces that individuals experience. With this idea in mind, Gillian Rhodes and colleagues (2003) performed experiments that showed that exposure to distorted faces can shift, at least temporarily, what perceivers perceive as average and consequently what they judge as attractive. They thus suggest that "people's preferences can be rapidly calibrated to match whatever physical characteristics are typical of the population of faces that they see" (p. 558). This suggests one way in which increasing interracial contact may reduce phenotypic devaluation: by increasing exposure to black faces, it may effect a change in what non-blacks perceive as average and so what they perceive as attractive.

Rhodes and colleagues' study may seem to offer powerful evidence in favour of a possibility that my argument denies, but there is a very good reason to question it. The theory on which it is based would seem to predict that persons living in societies with same-race majorities should have a same-race preference in judgements of attractiveness, since they will be mostly exposed to persons of the same race. But considerable evidence confounds this prediction. There is no general same-race

preference in judgements of attractiveness (Burke et al., 2013; Lewis, 2011; Rhodes et al., 2001). Participants in several studies exhibit an *other-race* preference, including individuals who live in societies where there are few other-race individuals.

#### **V.2**

There are at least three objections that my argument will prompt. First, it may be argued that, contrary to what I have suggested, phenotypic devaluation is in fact closely linked to stigmatization such that to the extent that the latter declines, so too will the former. The suggestion is that negative aesthetic judgements about blacks depend on their stigmatized status, so that if blacks are no longer stigmatized, characteristically black phenotypic traits will no longer be negatively evaluated. Recent trends in popular culture may be offered in support of it. One historian, for example, has pointed out that since the 1980s an African American has won the Miss USA pageant once every three years (Watson, 2010). And, in his recent book on black aesthetics, Paul Taylor (2016) suggests that there appears to be a narrowing of what he calls the "beauty gap" between African Americans and White Americans (Ch. 4; Taylor does recognize that things are more complicated than they appear). It may be argued that these changes reflect the fact that blacks are less stigmatized than they used to be.

In order to properly evaluate this argument, we need to be clear about how we should understand it. Recall first the distinction between pure and impure aesthetic judgements; aesthetic judgements, I suggested, can be infected by non-aesthetic impurities that taint them. Now if the objection is understood as claiming that eliminating or reducing black stigmatization will lead to revisions of judgements of black attractiveness because these are currently infected by the low social status of blacks and so are not pure, there may be something to it. It is not hard to believe that the negative aesthetic evaluation of a low social status group has been influenced by the group's low status. But the argument goes beyond this to claim that phenotypic devaluation is entirely a function of non-aesthetic impurities, and this is harder to believe. We do not generally think that social status is all that there is to judgements of attractiveness. And, in the case of blacks, many people can and do find many individual blacks to be physically attractive despite their low social status (though this point, by itself, is admittedly far from conclusive). And judgements of black unattractiveness seem to have been widely shared prior to the various ways in which stigmatization would later manifest itself (Barker, 1978; Jordan, 1968, pp. 8-11). But the main thing that I will say here is that the burden of argument should surely fall on those who would reject the moderate view that social status can influence judgements of attractiveness and embrace the more extreme view that such judgements are entirely a function of such factors. So, until that burden is fulfilled, I conclude that though it is likely that the history of stigmatization has influenced phenotypic devaluation, it is doubtful that it is wholly a function of it.

<sup>&</sup>lt;sup>49</sup> For example, Stephanie Camp (2015) has claimed that "It was within the context of modern slavery in the Americas that African and black bodies came to be seen as singularly and uniformly ugly" (p. 681).

<sup>&</sup>lt;sup>50</sup> Also worth noting is that, in 2019, for the first time, women of African descent held the Miss USA, Miss Teen USA, and Miss America titles at the same time (see Zaveri, 2019). However, two of the three women are bi-racial.

The evidence that I suggested may be offered in support of the objection should also be questioned. First, to the extent that there is indeed a narrowing of the so-called 'beauty gap,' it may more reflect changes in black grooming practices (such as hair processing techniques) and black phenotypes (through reproductive exogamy) than anything else. And, though it can be argued that the popularity of certain curvier African American female entertainers shows that there has been a change in body ideals, it is more likely that African Americans have created an alternative ideal that competes with, but is still subordinate to, a still-dominant mainstream ideal that valorizes thinness. Again, the case of Brazil is instructive. In Brazil, there is a racially mixed ideal defined in part by a curvier body shape. But this exists alongside a thinner ideal that some maintain is "associate[d] with a more distinguished taste" (Jarrín, 2015, p. 547). More importantly, it is far from clear that there has been any change at all for ideals of facial beauty, with unmixed blackness still considered the most unattractive.<sup>51</sup> As Donna Goldstein (1999) explains about the Brazilian context, "The generally accepted equation is that a particular combination of white and black characteristics creates mulata beauty, but it is white characteristics alone that can also qualify in another (higher) category of beauty. Purely African characteristics with no mixture of white characteristics are considered ugly" (p. 569).

Consider now a second objection. As the reader will have noticed, I have repeatedly used Brazil as a real-life example of the potential self-worth harms of integration in circumstances of stigmatization and phenotypic devaluation. It may be claimed that this use of Brazil is unfair, for it suggests that in evaluating the permissibility and wisdom of integration its benefits have to be weighed against self-worth harms of an order of magnitude that all but assures the conclusion that integration must be rejected. In other words, it may be argued that the self-worth harms of integration can outweigh the benefits of integration only if something like a worst-case scenario obtains, but this is very unlikely to be the case at this point. We have simply progressed too far for another Brazil to be a real possibility.

This objection invites us to consider the facts about anti-black racism and phenotypic devaluation that obtain in the societies that concern us. I have already indicated that no detailed examination of this issue can be undertaken here. Still, I want to concede that Brazil is indeed something close to a worst-case scenario that may be unlikely to be *fully* repeated elsewhere. But the self-worth harms of integration do not have to reach Brazil-levels for them to outweigh the (realistic) benefits of integration, and if what I have argued about their absolute overriding weight compared to these benefits is right, likely harms will outweigh them.

Turn now to a third and final objection. It may be argued that in rejecting integration in order to protect black self-worth, I ensure that blacks will continue to suffer from the stigmatizing effects of a segregated racial inequality, which itself imposes its own self-worth harms. After all, it will be argued, I do not deny that such inequality is stigmatizing, though I reject the necessary means for ending it. Now I concede that segregated racial inequality can express stigmatizing messages, and that integration

<sup>&</sup>lt;sup>51</sup> According to Alvaro Jarrín (2015), the ideal of a miscegenated Brazilian beauty "relies on the assumption that any desirable racialized characteristics can only be located in the buttocks and breasts, while for facial characteristics the desirable outcome is always one of whitening a person's features" (p. 543).

can help reduce this. But I do not believe that racial inequality cannot be ended or reduced without racial integration. Part of my thinking here is that I believe that there are a number of areas where action neutral with regard to integration might be taken to improve black opportunities. For example, there are ways of improving the education of black children that do not involve integrating their schools (Robbie & Fryer, 2011; Tough, 2008). But a bigger part of it is that I believe that we should not underestimate what a committed and determined black population can achieve despite the significant obstacles that it faces. Crucial here will be an honest reckoning with the legacies of past oppression. Also likely of some importance is some significant degree of class integration within black communities. Space constraints limit how much detail can be provided here, but I do not believe that racial inequality is inevitable. Moreover, without it, segregation — especially if voluntary — need not be stigmatizing. Si

#### **V.3**

The argument thus far suggests a future where the increasing interracial contact that integration involves reduces black stigmatization, but not phenotypic devaluation. Things may in fact be more complicated, but not for the better. For it is quite possible that insofar as stigmatization is reduced, this will increase the saliency of phenotypic variation among blacks, and heighten the devaluation of those blacks who rate high in phenotypic stereotypicality.

At this point, we must revisit an important assumption with which we have been working — something that will require conceding an important limitation of the overall argument. I have been assuming that the black population whose integration we are considering is a relatively homogenous group phenotypically. We must now relax this assumption. Individual blacks in fact differ significantly in the extent to which they are stereotypical phenotypically: some blacks are darker than others, have wider noses, and so on. There is reason to think that blacks who are more stereotypical will be more subject to phenotypic devaluation and so relational devaluation.

<sup>&</sup>lt;sup>52</sup> Racial oppression and racially hostile environments of other kinds can over time engender adaptations that, while functional in some respects, are dysfunctional in others. For example, an environment that threatens black self-esteem might give rise to norms that, while functional in protecting black self-esteem, are dysfunctional in not allowing blacks to fully take advantage of educational opportunities. Intelligently addressing such dysfunctions are essential if blacks are to significantly reduce racial inequality despite segregation. Unfortunately, many blacks reject out of hand any suggestion that community norms could be dysfunctional in any way, often accusing those who voice such views of blaming the victim. But as my description itself suggests, such views may be advanced while placing the blame squarely on the shoulders of those responsible for creating the hostile environment. Anderson (2010, Ch. 4, esp. pp. 75–84) has a good discussion of these issues, but I believe that there is more that blacks can do on their own on these fronts than she seems to think possible.

<sup>&</sup>lt;sup>53</sup> Here is an additional objection that I believe has already been answered by the argument developed, but which may be worth raising and responding to directly. The objection is raised by an anonymous referee, who asks (paraphrasing only slightly) 'If egalitarian anti-racist goals have been achieved, who cares whether or not you're liked?' The answer is: blacks may care. And if the blacks in question are assimilated, they almost certainly will care. What pre-empts such concern among some blacks at present is the group boundaries that I have argued will be threatened in an integrated society.

For example, in three studies Michelle Hebl and colleagues (2012) found that "being perceived as more physically stereotypical was associated with having a smaller group of outgroup friends and fewer outgroup social interactions" (p. 1333). In one particularly important study, they altered photographs of moderately stereotypical blacks (one male and one female) to make them more and less stereotypical, and used these to create Facebook profiles, which were then used to send out friendship requests to randomly selected profiles. They found that "an overture of friendship from a Black person to a non-Black person was less likely to be accepted when the friendship requester looked more (vs. less) stereotypically Black. Further, even among those requests that were accepted, more time passed before acceptance when the requester looked more (vs. less) stereotypically Black" (p. 1331).<sup>54</sup>

It may well be that the best kind of integration we can realistically hope for is one that will differentially incorporate blacks in the way suggested by Hebl and colleagues' study. Lighter-skinned blacks and blacks with other less stereotypical features (e.g., narrower noses) may come to enjoy the full benefits of integration, while more stereotypical blacks are denied the same. Thus the fact that blacks as a group are characterized by significant phenotypic variability presents a problem for integrationists. But it also presents a problem for my overall argument, for the individual arguments that I have made apply entirely at the level of the individual, and yet individual blacks vary in their stereotypicality. So, for all that I have argued so far, it is not clear why less stereotypical blacks should not integrate. As members of a still-stigmatized group, to be sure, they will experience some relational devaluation rooted in stigmatization, but over time — as the integration process unfolds — this may be expected to decrease. Moreover, they would not be relationally devalued because of phenotypic devaluation. As a result, integration for them would not inflict, to the same degree, the self-worth harms that we have discussed.

A different possibility must be considered. It is possible that the integration of less stereotypical blacks, increasingly on terms of equality, will pave the way for the later integration of more stereotypical blacks. It may be argued that less stereotypical blacks might constitute a kind of 'gateway' group, exposure to which will improve general anti-black attitudes, the benefits of which will redound to all blacks. In a series of studies on this issue, Aharon Levy and colleagues (2019) found that the presence of bi-racial individuals reduced prejudice through the reduction of symbolic threat. But even if it is true that less stereotypical blacks can reduce the prejudice directed at more stereotypical blacks, it is a separate question whether they can reduce their phenotypic devaluation. I see no reason to draw this inference, based on this study.

So, if such gateway hopes should not be relied on, how do I propose dealing with the problem of phenotypic variation? At this point, I turn to an ideal of black solidarity. Solidarity among the members of a group can be effective in achieving important goals (such as justice for the group). But solidarity may also be valuable because

<sup>&</sup>lt;sup>54</sup> There is a question here about whether these results reflect phenotypic devaluation or stigmatization: are more stereotypical blacks more subject to social rejection because they are phenotypically devalued, or is it because they are more likely to be stereotyped? As I explain below, perceiving subjects are more likely to draw stereotypic inferences about them. All things considered, I think that it is more likely that the results reflect phenotypic devaluation, but I won't attempt to show this here.

of the value of the goods that it embodies (e.g., mutual trust, sense of special concern, and shared fate). Because this is so, it may motivate a principled stance toward some more or less permanent adversity, faced by some but not all members of the group, where there is no clear prospect in the foreseeable future that collective action will be effective in overcoming it. In such cases, solidarity may demand that more fortunate members of the group decline certain benefits that other members can't share. Acceptance of these benefits may be particularly destructive of solidarity. I believe that something like this is true for blacks. In particular, I suggest that solidarity demands that less stereotypical blacks decline opportunities for social inclusion that are denied to more stereotypical blacks. Included here are opportunities to integrate.<sup>55</sup>

To be sure, it is not always inconsistent with black solidarity (or rather, with an ideal of it worth defending) for less stereotypical blacks to accept benefits denied to other blacks. Take the economic domain. It would seem that a high-income earning business executive can keep her high income for herself even when her lesser stereotypicality was a factor in — even a necessary condition of — her career ascent. Phenotypic variation makes a difference not only with respect to phenotypic devaluation, but stigmatization as well: blacks who are more stereotypical are more likely to be subject to negative evaluation (e.g., judged as less intelligent) and stereotyping (Ben-Zeev et al., 2014; Hannon, 2015; Maddox, 2004).<sup>56</sup> Thus lighter-skinned African Americans have been found to earn higher wages than darker-skinned African Americans, with the wage gap widening as the skin colour of workers darken (Goldsmith et al., 2007). It also has an impact in the domain of electoral politics (Hochschild & Weaver, 2007, pp. 650-652). Enjoying success here too would seem to be consistent with black solidarity. But social situations, I claim, are different. Here the acceptance by less stereotypical blacks of benefits denied to more stereotypical blacks is uniquely destructive of black solidarity, and so of the goods that it embodies.

As this suggests, it is an implication of my argument that while some blacks should reject integration on self-protective grounds, some should reject it on solidarity grounds. Interestingly, on my view, blacks who should reject it on solidarity grounds may have stronger — more absolute — reasons to do so than those who should reject it on self-protection grounds. Say that a society has some stereotypical blacks, some less stereotypical blacks, and that there are benefits to integrating. My view is that the

<sup>&</sup>lt;sup>55</sup> With the long entrenchment of the one-drop rule in the US, and the way that over time it compelled racially mixed blacks to embrace blackness, as opposed to a separate identity, many African Americans already practice the kind of solidarity for which I am advocating here. But the one-drop rule is decidedly the exception, not the rule, and I surmise that outside of the US it would be difficult to find similar solidarity practices, save perhaps in places heavily influenced by African American culture. (Recall that I take my argument to have international application.) For discussion of the uniqueness of the one-drop rule, see (Davis, 1991). Now there is a qualification of my argument that should be mentioned. The obligation of solidarity for which I am advocating here does not apply to all those who qualify as 'black' by the one-drop rule. The considerations that lead me to this position are beyond the scope of the arguments of this paper, but I will say that it in part reflects my rejection of the rule itself.

<sup>&</sup>lt;sup>56</sup> These facts about differential black vulnerability to stigmatization cast serious doubt on Shelby's (2005) claim that the common experience of anti-black racism can be a firm foundation for black solidarity. Black solidarity should be founded on something that enables it to survive differential black experience with racism, and given that differential experience, 'the common experience of anti-black racism' cannot be it.

benefits of integration can be given more weight by those blacks who, for their part, would evaluate it from the perspective of its self-worth harms than those who would evaluate it from the perspective of the demands of solidarity.

So phenotypic variation limits my argument in that it requires recourse to an ideal of solidarity that cannot be fully defended here. But the very facts that give rise to this limitation also raise the prospect that the self-worth harms of integration may be more significant than I have so far suggested. An integrated society would presumably contain more interracial unions, and this will over time increase the number of blacks who are less stereotypical. It may be that once the population balance of blacks who are more and those who are less stereotypical shifts in favour of the latter, this will give more licence to open expressions of phenotypic devaluation of the former. Some of us may think that our societies have progressed beyond such open denigration, but once it is understood that the denigration is *feature*- and not *category*-based, old norms against category-based denigration may no longer seem to apply. Moreover, this denigration may seem different if it is done by persons who are also considered to be black.

#### **V.4**

Let us turn now, more briefly, to the second question about intergroup contact, which concerns whether it has any unintended and unwelcome side effects. The problem here is that research on interracial contact indicates, not surprisingly, that it is close cross-racial group friendships that are "especially effective in reducing prejudice and promoting positive intergroup outcomes" (Pettigrew & Tropp, 2011, p. 119). The idea is that it is such friendships that can reduce intergroup anxiety and trigger the affective processes that are so important to prejudice reduction.

This is concerning because, as we have seen, the problem with integration is that it will tend to dissolve group boundaries, and what effects this result are the assimilative pressures that I have claimed it will inevitably exert. Close personal friendships would seem to be a main conduit for these pressures. So it seems that the very thing that may be most effective in reducing prejudice is just what will most expose blacks to the self-worth harms of integration.

If this is right, it shows that if integration is to fully produce its claimed beneficial effects, it may require forms of interracial contact that will likely lead to psychological integration, which I have claimed significantly increases the risk of assimilation. This in turn suggests that integrationists are in a dilemma: either integration will fall short with respect to close personal relationships, thereby possibly avoid assimilation but risk failing to sufficiently reduce even stigmatization, or it will achieve the requisite level of close personal relationships, achieve de-stigmatization, but lead to assimilation, and so expose blacks — especially those who are more stereotypical — to selfworth harms rooted in phenotypic devaluation.

So, even if integration can be effective in reducing stigmatization, it is not only doubtful whether it can also reduce phenotypic devaluation, but it may in fact make it worse for some blacks. But something must also be said about the 'transition costs' — transition self-worth harms — that integration will involve, for de-stigmatization, even if successful over the long term, can't happen overnight.

The self-worth harms of relational devaluation rooted in stigmatization will not end as soon as the integration process begins. This is potentially quite significant, if one places great weight on avoiding these harms, as I do. Now if we add to this the uncertainty of the long-term effects of integration, on the one hand, and its very likely immediate self-worth harms, on the other, it would appear that integrationists want to risk sacrificing current and immediately succeeding generations in the name of a good whose attainment is far from guaranteed. For these reasons, I suggest that even if we put phenotypic devaluation aside, concerns about stigmatization alone seem to provide enough reason for blacks to reject integration.

#### VI.1

Discussions of the pros and cons of integration have not given enough attention to the issue of self-worth. In this paper I have argued that integration will likely have self-worth costs, and that these costs may be so significant that blacks should reject it on self-protective and solidarity grounds. It will, I argue, inevitably exert assimilative pressures, and assimilation may intensify the self-worth harms of stigmatization and phenotypic devaluation by leading blacks to internalize racial and/or phenotypic hierarchies. In making this argument, I have sought to stress the importance of considering phenotypic traits in thinking about integration. Unless we attend to the way that racial features differentially affect the black experience with stigmatization and especially phenotypic devaluation, I suggest, we may fail to notice what may be integration's greatest threat. The problem is that in conditions of phenotypic devaluation it may make blacks aspire, not to whiteness, but to a lesser kind of blackness. And we may fail to notice this problem if we focus exclusively on racial categories.<sup>57</sup>

Now what does my argument mean for specific forms of integration? The answer to this is less clear than it may be appear. To be sure, I certainly maintain that assimilation should be avoided, and the most vulnerable blacks in this regard would seem to be black children. But integration in which domains will lead to the assimilation of black children? Though I earlier singled out residential integration as posing a special threat, I do not believe that it makes black internalization of racial or phenotypic hierarchies inevitable. For certain individual blacks, a sufficiently robust sense of racial identity may be compatible with residential integration, even when paired with school integration — there may be a certain psychological orientation that can block some of the effects of residential integration. Crucial here are parental practices of racial socialization. The skilled use of such practices may protect the self-worth of even children raised in predominately white environments where blacks continue to be devalued, especially if blacks form a sufficient critical mass. But this should not generally be relied on. Some black parents may not have the requisite knowledge and skills to appropriately socialize their children; these must be learned, but we should not presume that all blacks have done so. This is especially true for black immigrants who have no history of living in multiracial societies. By contrast, the racial socialization of black children is less important in a context of residential segregation.

<sup>&</sup>lt;sup>57</sup> Some will want to make this point by saying that we should think in terms of colour, not race. But since colour is only one phenotypic trait, and probably not the most important, this isn't quite right.

At the same time, more innocuous-seeming forms of interracial contact may come with surprising risks, even in a context of residential segregation. The problem here has to do with the fact that, as Michael Merry (2013) has written, "one can live one's entire life within a segregated neighborhood and in a number of crucial ways still be integrated" (p. 170). For example, a person might live in a black neighbourhood while having a mostly white circle of friends. This kind of situation will exert assimilative pressures too. Thus, if my argument is right, it should be avoided.

In her book on integration, Stanley (2017) distinguishes between a temporary resistance to integration and a more permanent rejection of it (pp. 187–188). If my argument is right, it is true that it suggests a resistance that may be more than temporary: as long as phenotypic devaluation persists, blacks should resist integration. And yet my argument doesn't quite provide reason to favour a final rejection of it either. For if it becomes clear that phenotypic devaluation is not quite the problem that I believe it is, then one main reason for blacks to resist integration can be put aside. Until then, however, blacks should consider it off the table.

#### VI.2

Now it may be wondered why I should devote so much energy to arguing against something that I have already indicated has fallen out of favour among most African Americans. The primary answer has to do with something that I mentioned in the introduction: I intend my argument to have international application. I suspect that when we turn our attention to non-African American blacks — African Canadians, Black British people, and others — we will find that there are a great many who still find integration appealing.<sup>58</sup>

And yet if integration is more appealing to other blacks, they generally may be more at risk of suffering the self-worth harms I have described. I have suggested that blacks are differentially subject to phenotypic devaluation based on their stereotypicality — in particular, that more stereotypical blacks are more subject. If this is right, then integration will not inflict the same degree of harm on all blacks. But now notice that this is true not just at the level of the individual, but also at the level of the state. Different black nationalities differ in the extent of their general stereotypicality; African Americans, for example are less stereotypical than African Canadians, largely because most African Canadians are recent immigrants or the descendants of recent immigrants from countries with a less significant history of intermixing. If all of this is true, then the blacks who are most likely to still find integration appealing are just those who are most likely to be harmed by it.

Acknowledgements. Earlier versions of this paper were presented at York University, Queens University and Texas A&M University. I would like to thank the audiences for their questions and comments. I am also grateful to the six anonymous referees for their comments and to Jill Flohil for her copyediting. Special thanks to Louis Philippe-Hodgson for reading and providing comments on two earlier versions of this paper.

<sup>&</sup>lt;sup>58</sup> This has to do with characteristics of the source countries. For example, many Black British people are descendants of recent immigrants from the West Indies, and West Indians are pro-integration in outlook (see esp. Lowenthal, 1972).

## References

- Adams, M. (2006). Radical integration. California Law Review, 94(2), 261-311.
- Alba, R., Beck, B., & Sahin, D. B. (2018). The U.S. mainstream expands again. Journal of Ethnic and Migration Studies, 44(1), 99–117.
- Anderson, E. (1999). What is the point of equality? Ethics, 109(2), 287-337.
- Anderson, E. (2010). The imperative of integration. Princeton University Press.
- Appiah, K. A. (1996). Race, culture, identity: Misunderstood connections. In K. A. Appiah & A. Gutmann (Eds.), *Color conscious: The political morality of race* (pp. 30–105). Princeton University Press.
- Barker, A. J. (1978). The African link: British attitudes to the negro in the era of the Atlantic slave trade, 1550–1807. Frank Cass.
- Ben-Zeev, A., Dennehy, T. C., Goodrich, R. I., Kolarik, B. S., & Geisler, M. W. (2014). When an "educated" black man becomes lighter in the mind's eye: Evidence for a skin tone memory bias. *SAGE Open*, (4)1, 1–9
- Bird, C. (2010). Self-respect and respect for others. European Journal of Philosophy, 18(1), 17-40.
- Blum, L. (2002). "I'm not a racist, but": The moral quandary of race. Cornell University Press.
- Boxill, B. R. (1992). Blacks and social justice (rev. ed.). Rowman & Littlefield Publishers.
- Brooks, R. L. (1996). Integration or separation: A strategy for racial equality. Harvard University Press. Brown v. Board of Education of Topeka, 347 U.S. 483 (1954).
- Burke, D., Nolan, C., Hayward, W. G., Russell, R., & Sulikowski, D. (2013). Is there an own-race preference in attractiveness? *Evolutionary Psychology*, 11(4), 855–872.
- Camp, S. M. H. (2015). Black is beautiful: An American history. Journal of Southern History, 81(3), 675–690.
- Cashin, S. (2004). The failures of integration: How race and class are undermining the American dream. Public Affairs.
- Cashin, S. (2017). Loving: Interracial intimacy and the threat to white supremacy. Beacon Press.
- Clark, K. B., & Clark, M. P. (1947). Racial identification and preference in negro children. In T. M. Newcomb & E. L. Hartley (Eds.), *Readings in social psychology* (pp. 169–178). Holt, Rinehart, and Winston.
- Corcoran, K., Crusius, J., & Mussweiler, T. (2011). Social comparison: Motives, standards, and mechanisms. In D. Chadee (Ed.), *Theories in social psychology* (pp. 119–139). Wiley-Blackwell.
- Crocker, J., & Major, B (1989). Social stigma and self-esteem: The self-protective properties of stigma. *Psychological Review*, 96(4), 608–630.
- Crocker, J., Major, B., & Steele, C. (1998). Social stigma. In D. T. Gilbert, S. T. Fiske, & G. Lindzey (Eds.), Handbook of social psychology (4th ed., vol. 1, pp. 504–553). McGraw-Hill.
- Cross, W.E., Jr. (1991). Shades of black: Diversity of African American identity. Temple University Press. Darwall, S. L. (1977). Two kinds of respect. Ethics, 88(1), 36–49.
- Davis, F. J. (1991). Who is black? One nation's definition. Pennsylvania State University Press.
- Dillon, R. S. (1992). How to lose your self-respect. American Philosophical Quarterly 29(2), 125-139.
- Dillon, R. S. (2013). Self-respect and self-esteem. In H. LaFollette (Ed.), The international encyclopedia of ethics, 8, (4772–4782). Wiley-Blackwell Publishing.
- Dillon, R. S. (2018). Respect. In E. N. Zalta (Ed.), *The Stanford encyclopedia of philosophy* (Spring 2018 ed.). Stanford University. https://plato.stanford.edu/archives/spr2018/entries/respect
- Dovidio, J. F., Love, A., Schellhaas, F. M. H., & Hewstone, M. (2017). Reducing intergroup bias through intergroup contact: Twenty years of progress and future directions. *Group Processes and Intergroup Relations*, 6(1), 606–620.
- Ellemers, N., & Haslam, S. A. (2012). Social identity theory. In P. A. M. van Lange, A. W. Kruglanski, & E. T. Higgins (Eds.), *Handbook of theories of social psychology* (vol. ii, pp. 379–398). Sage.
- Fanon, F. (1967). Black skins, white masks (C. L. Markmann, Trans.). Grove Press.
- Flew, A. (1990). Three concepts of racism. Encounter, 75, 63-66.
- Fourie, C. (2012). What is social equality? An analysis of status equality as a strongly egalitarian ideal. *Res Publica*, 18(2), 107–126.
- Fourie, C. (2015). To praise and to scorn: The problem of inequalities of esteem for social egalitarianism. In C. Fourie, F. Schuppert, & I. Wallimann-Helmer (Eds.), *Social equality: On what it means to be equals* (pp. 87–106). Oxford University Press.

- Frevert, T. K., & Walker, L. S. (2014). Physical attractiveness and social status. *Sociology Compass*, 8(3), 313–323.
- Gans, H. J. (1999). The possibility of a new racial hierarchy in the twenty-first-century United States. In M. Lamont (Ed.), The cultural territories of race: Black and white boundaries (pp. 371–390). University of Chicago Press.
- Garcia, J. L. A. (1996). The heart of racism. Journal of Social Philosophy, 27(1), 5-45.
- Garcia, J. L. A. (1997). Current conceptions of racism. Journal of Social Philosophy, 28(2), 5-42.
- Goffman, E. (1963). Stigma: Notes on the management of spoiled identity. Simon & Schuster.
- Goldsmith, A. H., Hamilton, D., & Darity, Jr., W. (2007). From dark to light: Skin color and wages among African-Americans. *Journal of Human Resources*, 42(4), 701–738.
- Goldstein, D. (1999). "Interracial" sex and racial democracy in Brazil: Twin concepts? American Anthropologist, 101(3), 563–578.
- Gordon, D. (2019). Body beautiful: Comparative meanings of beauty in Brazil, South Africa and Jamaica. In C. Liebelt, S. Böllinger, & U. Vierke (Eds.), *Beauty and norm: Debating standardization in bodily beauty* (pp. 215–244). Palgrave Macmillan.
- Hannon, L. (2015). White colorism. Social Currents, 2(1), 13-21.
- Harter, S. (1993). Visions of self: Beyond the me in the mirror. In J. E. Jacobs (Ed.), Nebraska symposium on motivation, 1992 Volume 40: Developmental perspectives on motivation (pp. 99–144). University of Nebraska Press.
- Hebl, M. R., Williams, M. J., Sundermann, J. M., Kell, H. J., & Davies, P. G. (2012). Selectively friending: Racial stereotypicality and social rejection. *Journal of Experimental Social Psychology*, 48(6), 1329–1335.
- Hill, T. (1995a). Servility and self-respect. Reprinted in R. S. Dillon (Ed.), *Dignity, character, and self-respect* (pp. 76–93). Routledge.
- Hill, T. (1995b). Self-respect reconsidered. Reprinted in R. S. Dillon (Ed.), *Dignity, character, and self-respect* (pp. 117–125). Routledge.
- Hobbs, A. (2014). A chosen exile: A history of racial passing in American Life. Harvard University Press.
- Hochschild, J. L., & Weaver, V. (2007). The skin color paradox and the American racial order. *Social Forces*, 86(2), 643–670.
- Holberg, E. (2017). Kant, oppression, and the possibility of non-culpable failures to respect oneself. Southern Journal of Philosophy, 55(3), 285–305.
- Holden, C. J., Vrabel, J. K., & Zeigler-Hill, V. (2016). Self-esteem as a status-tracking mechanism. In T. K. Shackelford & V. A. Weekes-Shackelford (Eds.), Encyclopedia of evolutionary psychological science. Springer. https://doi.org/10.1007/978-3-319-16999-6\_1443-1
- Hordge-Freeman, E. (2015). The color of love: Racial features, stigma, and socialization in black Brazilian families. University of Texas Press.
- Hughes, M., Kiecolt, K. J., Keith, V. M., & Demo, D. H. (2015). Racial identity and well-being among African Americans. Social Psychology Quarterly, 78(1), 25–48.
- Jarrín, A. (2015). Towards a biopolitics of beauty: Eugenics, aesthetic hierarchies and plastic surgery in Brazil. Journal of Latin American Cultural Studies, 24(4), 535–552.
- Jordan, W. D. (1968). Black over white: American attitudes toward the negro, 1550–1812. University of North Carolina Press.
- Kennedy, R. (2003). Interracial intimacies: Sex, marriage, identity and adoption. Pantheon.
- Kitchner, P. (2007). Does race have a future? Philosophy and Public Affairs, 35(4), 293-317.
- Kymlicka, W. (1998). Finding our way: Rethinking ethnocultural relations in Canada. Oxford University Press.
- Leary, M. R. (2005). Sociometer theory and the pursuit of relational value: Getting to the root of self-esteem. *European Review of Social Psychology*, 16(1), 75–111.
- Leary, M. R., & Allen, A. B. (2011). Belonging motivation: Establishing, maintaining, and repairing relational value. In D. Dunning (Ed.), *Social motivation* (pp. 37–55). Psychology Press.
- Leary, M. R., & Baumeister, R. F. (2000). The nature and function of self-esteem: Sociometer theory. In M. P. Zanna (Ed.), *Advances in experimental social psychology*, 32, (pp. 1-62). Academic Press.
- Leary, M. R., & Terry, M. L. (2013). Self-evaluation and self-esteem. In D. Carlston (Ed.), *The Oxford handbook of social cognition* (pp. 534–547). Oxford University Press.
- Lee, J., & Bean, F. D. (2010). *The diversity paradox: Immigration and the color line in 21<sup>st</sup> century America*. Russell Sage Foundation.

- Levy, A., Halperin, E., van Zomeren, M., & Saguy, T. (2019). Inter-racial gateways: The potential of biracials to reduce threat and prejudice in inter-racial dynamics. *Race and Social Problems*, 11(2), 119–132.
- Lewis, M. B. (2011). Who is the fairest of them all? Race, attractiveness and skin color sexual dimorphism. *Personality and Individual Differences*, 50(2), 159–162.
- Liu, X. (2018). Discrimination and lookism. In K. Lippert-Rasmussen (Ed.), *The Routledge handbook of the ethics of discrimination* (pp. 276–286). Routledge.
- Loury, G. C. (2002). The anatomy of racial inequality. Harvard University Press.
- Lowenthal, D. (1972). West Indian societies. Oxford University Press.
- Maddox, K. B. (2004). Perspectives on racial phenotypicality bias. Personality and Social Psychology Review, 8(4), 383–401.
- Major, B., McCoy, S. K., Kaiser, C. R., & Quinton, W. J. (2003). Prejudice and self-esteem: A transactional model. European Review of Social Psychology, 14(1), 77–104.
- Makkar, J. K., & Strube, M. J. (1995). Black women's self-perceptions of attractiveness following exposure to white versus black beauty standards: The moderating role of racial identity and self-esteem. *Journal of Applied Social Psychology*, 25(17), 1547–1567.
- Mallon, R. (2017). Racial identity, racial ontology, and racial norms. In N. Zack (Ed.), The Oxford handbook on philosophy and race (pp. 392–401). Oxford University Press.
- Margalit, A. (1996). The decent society (N. Goldblum, Trans.). Harvard University Press.
- Massey, S. J. (1983). Kant on self-respect. Journal of the History of Philosophy, 21(1), 57-73.
- Matthew, D. C. (2017). Racial injustice, racial discrimination and racism: How are they related? *Social Theory and Practice*, 43(4), 885–914.
- McGary, H. (1999). Racial integration and racial separatism: Conceptual clarifications. Reprinted in H. McGary (Ed.), *Race and social justice* (pp. 43–61). Blackwell Publishers.
- McPherson, L. K. (2015). Deflating "race." Journal of the American Philosophical Association, 1(4), 674–693.
- Mensah, J. (2010). Black Canadians: History, experience, social conditions (2<sup>nd</sup> ed.). Fernwood Publishing.
- Merry, M. S. (2013). Equality, citizenship, and segregation: A defense of separation. Palgrave Macmillan. Mitchell, M., & Wells, M. (2018). Race, romantic attraction, and dating. Ethical Theory and Moral Practice,
- Mitchell, M., & Wells, M. (2018). Race, romantic attraction, and dating. *Ethical Theory and Moral Practice* 21(4), 945–961.
- Neal, A. M., & Wilson, M. L. (1989). The role of skin color and features in the black community: Implication for black women and therapy. *Clinical Psychology Review*, 9(3), 323–333.
- Ogbu, J. U., & Simons, H. D. (1998). Voluntary and involuntary minorities: A cultural-ecological theory of school performance with some implications for education. *Anthropology and Education Quarterly*, 29(2), 155–188.
- Orth, O., & Robbins, R. W. (2014). The development of self-esteem. Current Directions in Psychological Science, 23(5), 381-387.
- Parmer, T., Arnold, M. S., Natt, T., & Janson, C. (2004). Physical attractiveness as a process of internalized oppression and multigenerational transmission in African American families. *The Family Journal*, 12(3), 230–242.
- Patterson, O. (1997). The ordeal of integration. Citivas/Counterpoint.
- Pettigrew, T. F., & Tropp, L. R. (2011). When groups meet: The dynamics of intergroup contact. Psychology Press
- Philips, M. (1984). Racist acts and racist humor. Canadian Journal of Philosophy, 14(1), 75-96.
- Rawls, J. (1999). A theory of justice (2nd ed.). Harvard University Press.
- Reece, R. L. (2016). What are you mixed with: The effect of multiracial identification on perceived attractiveness. *Review of Black Political Economy*, 43(2), 139–147.
- Rhodes, G. (2006). The evolutionary psychology of facial beauty. *Annual Review of Psychology*, 57(1), 199–226.
- Rhodes, G., Jeffery, L., Watson, T. L., Clifford, C. W. G., & Nakayama, K. (2003). Fitting the mind to the world: Face adaptation and attractiveness aftereffects. *Psychological Science*, 14(6), 558–566.
- Rhodes, G., Yoshikawa, S., Clark, A., Lee, K., McKay, R., & Akamatsu, S. (2001). Attractiveness of facial averageness and symmetry in non-western cultures: In search of biologically based standards of beauty. *Perception*, 30(5), 611–625.
- Robbie, W., & Fryer, Jr., R. G. (2011). Are high-quality schools enough to increase achievement among the poor? Evidence from Harlem Children's Zone. *American Economic Journal: Applied Economics*, 3(3), 158–187.

- Roithmayr, D. (2014). Reproducing racism: How everyday choices lock in white advantage. New York University Press.
- Runciman, W. G. (1967). "Social" equality. The Philosophical Quarterly, 17(68), 221-230.
- Russell, K., Wilson, M., & Ronald Hall, R. (2013). The color complex: The politics of skin color in a new millennium (rev. ed.). Anchor Books.
- Ryabov, I. (2019). How much does physical attractiveness matter for blacks? Linking skin color, physical attractiveness, and black status attainment. Race and Social Problems, 11(1), 68–79.
- Sachs, D. (1981). How to distinguish self-respect from self-esteem. Philosophy and Public Affairs, 10(4), 346–360.
- Scanlon, T. M. (2003). The diversity of objections to inequality. Reprinted in T. M. Scanlon (Ed.), *The difficulty of tolerance: Essays in political philosophy* (pp. 202–218). Cambridge University Press.
- Scanlon, T. M. (2018). Why does inequality matter? Oxford University Press.
- Scheffler, S. (2015). The practice of equality. In C. Fourie, F. Schuppert, & I. Wallimann-Helmer (Eds.), Social equality: On what it means to be equals (pp. 21–44). Oxford University Press.
- Schemmel, C. (2019). Real self-respect and its social bases. Canadian Journal of Philosophy, 49(5), 628-651.
- Sears, D. O. (2015). The American color line and black exceptionalism. In J. P. Forgas, K. Fiedler, & W. D. Crano (Eds.), Social psychology and politics (pp. 321–336). Psychology Press.
- Shelby, T. (2005). We who are dark: The philosophical foundations of black solidarity. Harvard University Press.
- Shelby, T. (2012). Race. In D. Estlund (Ed.), The Oxford handbook of political philosophy (pp. 336–353). Oxford University Press.
- Shelby, T. (2016). Dark ghettos: Injustice, dissent, and reform. Harvard University Press.
- Singer, M. G. (1959). On duties to oneself. Ethics, 69(3), 202-205.
- Spencer, Q. (2012). What biological racial realism should mean. Philosophical Studies, 159(2), 181-204.
- Spencer, Q. (2019). How to be a biological racial realist. In J. Glasgow, S. Haslanger, C. Jeffers, & Q. Spencer, What is race? (pp. 73–110). Oxford University Press.
- Stanley, S. A. (2017). An impossible dream? Racial integration in the United States. Oxford University Press. Stark, C. A. (1997). The rationality of valuing oneself: A critique of Kant on self-respect. Journal of the
- Stark, C. A. (1997). The rationality of valuing oneself: A critique of Kant on self-respect. *Journal of the History of Philosophy*, 35(1), 65–82.
- Stubblefield, A. (2005). Ethics along the color line. Cornell University Press.
- Tajfel, H., & Turner, J. C. (2004). The social identity theory of intergroup behavior. Reprinted in J. T. Jost & J. Sidanius (Eds.), *Political psychology: Key readings* (pp. 276–293). Psychology Press.
- Taylor, P. C. (2016). Black is beautiful: A philosophy of black aesthetics. Wiley-Blackwell.
- Thomas, L. (1978). Rawlsian self-respect and the black consciousness movement. *The Philosophical Forum*, 9(2-3), 303-314.
- Thomas, L. (1995). Self-respect: Theory and practice. Reprinted in R. S. Dillon (Ed.), Dignity, character, and self-respect (pp. 251–270). Routledge.
- Tough, P. (2008). Whatever it takes: Geoffrey Canada's quest to change Harlem and America. Houghton Mifflin.
- Ture, K., & Hamilton, C. V. (1992). Black power: The politics of liberation in America. Vintage Books. (Original work published 1967.)
- Twenge, J. M., & Crocker, J. (2002). Race, ethnicity, and self-esteem: Meta-analyses comparing whites, blacks, Hispanics, Asians, and American Indians and comment on Gray-Little and Hafdahl (2000). Psychological Bulletin, 128(3), 371–408.
- Twine, F. W. (1998). Racism in a racial democracy: The maintenance of white supremacy in Brazil. Rutgers University Press.
- Wade, P. (1993). Blackness and race mixture: The dynamics of racial identity in Colombia. Johns Hopkins University Press.
- Wallace, H. M., & Tice, D. M. (2012). Reflected appraisal through a 21st-century looking glass. In M. R. Leary & J. P. Tangney (Eds.), Handbook of self and identity (2<sup>nd</sup> ed., pp. 124–140). Guilford Publications.
- Warren, J. W., & Twine, F. W. (1997). White Americans, the new minority? Non-blacks and the ever-expanding boundaries of whiteness. *Journal of Black Studies*, 28(2), 200–218.
- Wasserstrom, R. (1977). Racism, sexism, and preferential treatment: An approach to the topics. *UCLA Law Review*, 24(3), 581–615.

- Watson, E. (2010, February 21). The racial politics of Miss America. https://www.blackpast.org/african-american-history/racial-politics-miss-america/
- West, C. (2002). Prophecy deliverance! An Afro-American revolutionary Christianity (20<sup>th</sup> anniversary ed.). Westminster John Knox Press. (Original work published 1982.)
- Winks, R. W. (1997). *The blacks in Canada: A history* (2<sup>nd</sup> ed.). McGill-Queen's University Press. (Original work published 1971.)
- Yancey, G. (2003). Who is white? Latinos, Asians, and the new black/nonblack divide. Lynne Rienner.
- Zaveri, M. (2019, May 5). A first for U.S. pageants: Black women hold all 3 major titles at the same time. The New York Times, A16.

Cite this article: Matthew, D. C. (2023). Racial Integration and the Problem of Relational Devaluation. *Dialogue* 62(1), 3–45. https://doi.org/10.1017/S0012217321000251